

LUND UNIVERSITY

Analysis of key challenges to implementation of FEFO in perishable food supply chain

Kandasamy, Jayakrishna ; Vimal, K.E.K. ; Singh, Aditya Pratap ; Magnani, Aaryan ; Gokhale, Ameya; Jagtap, Sandeep

Published in: Journal of Agriculture and Food Research

DOI: 10.1016/j.jafr.2025.101848

2025

Document Version: Publisher's PDF, also known as Version of record

Link to publication

Citation for published version (APA): Kandasamy, J., Vimal, K. E. K., Singh, A. P., Magnani, A., Gokhale, A., & Jagtap, S. (2025). Analysis of key challenges to implementation of FEFO in perishable food supply chain. *Journal of Agriculture and Food Research*, Article 101848. https://doi.org/10.1016/j.jafr.2025.101848

Total number of authors: 6

Creative Commons License: CC BY

General rights

Unless other specific re-use rights are stated the following general rights apply: Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

· Users may download and print one copy of any publication from the public portal for the purpose of private study

or research.
You may not further distribute the material or use it for any profit-making activity or commercial gain

· You may freely distribute the URL identifying the publication in the public portal

Read more about Creative commons licenses: https://creativecommons.org/licenses/

Take down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

LUND UNIVERSITY

PO Box 117 221 00 Lund +46 46-222 00 00



Contents lists available at ScienceDirect

Journal of Agriculture and Food Research



journal homepage: www.sciencedirect.com/journal/journal-of-agriculture-and-food-research

Analysis of key challenges to implementation of FEFO in perishable food supply chain

Jayakrishna Kandasamy^a, K.E.K. Vimal^b, Aditya Pratap Singh^a, Aaryan Magnani^a, Ameya Gokhale^a, Sandeep Jagtap^{c,d,*}

^a School of Mechanical Engineering, Vellore Institute of Technology, Vellore, Tamil Nadu, India

^b Department of Production Engineering, National Institute of Technology, Tiruchirappalli, Tamil Nadu, India

^c Division of Engineering Logistics, Lund University, Lund, Sweden

^d Sustainable Manufacturing Systems Centre, Cranfield University, UK

ARTICLE INFO

Keywords: First-in first-out First expired first out Inventory management systems Perishable food supply chain Total interpretive structural modelling Decision-making trial and evaluation laboratory

ABSTRACT

Implementing FEFO practices has become essential for organizations globally to minimize spoilage, enhance inventory turnover, and ensure compliance with health and safety standards. To aid stakeholders in effectively adopting FEFO, it is crucial to identify and address the challenges involved in its implementation. Through an extensive literature review using the Preferred Reporting Items for Systematic reviews and Meta-Analyses (PRISMA) methodology and insights from industry experts, this study identifies thirteen core challenges that hinder FEFO adoption. PRISMA methodology was used to systematically organize the existing literature for the purpose of this study. Using tools like Decision Making Trial and Evaluation Laboratory (DEMATEL) and Total Interpretive Structural Modelling (TISM), the challenges were examined and ranked according to their interdependencies, providing insights into the cause-effect relationships among them. After applying DEMATEL, an alpha threshold value of 0.368 revealed that challenges in effective storage management are the primary barrier in implementing FEFO practices. With level partitioning, this challenge emerged as the most significant, forming the foundation for a roadmap designed to assist stakeholders. The findings from this study offer managers actionable insights for implementing effective FEFO techniques within their organizations. The study's novelty lies in its combination of DEMATEL and TISM methodologies, along with a roadmap that highlights strategic and policy-focused recommendations to support efficient FEFO adoption and the systematic study of challenges preventing effective FEFO adoption. This paper aids implementation of FEFO for better inventory control and management, reduced wastage and greater efficiency. The paper also effectively outlines and analyses the order of importance of challenges in FEFO implementation and their interdependence.

1. Introduction

The Perishable Food Supply Chain (PFSC) faces unique challenges where the shelf-life of products has an impact on both the economic viability and safety of food items. The shelf life of items in the perishable food industry directly affects factors such as safety and economic viability [1]. To minimize waste and ensure timely distribution of goods, effective inventory management is required. Earlier First-In, First-Out (FIFO) method was used, an inventory management method where the oldest items are used or sold first before newer items [2]. It is often used to manage stock rotation and ensure that older products are used or sold first. It keeps accurate chronological flow of goods incoming and outgoing and is commonly applied in industries dealing with perishable goods, such as dairy, meat, poultry, fruits, vegetables, pharmaceuticals etc. One inventory management approach that fulfils the criteria of both safety and minimal waste is the First Expired, First Out (FEFO) method that aims to reduce health hazards associated with consumption of expired goods [3]. The FEFO method works by dispatching items that are closest to expiry while still being safe for use unlike the (FIFO) method where dispatch of goods is solely decided by the order in which the items are received. FIFO leads to higher food waste as oldest item may not be closest to expiry [4]. It requires careful organization to ensure older items are at the front, leading to disorganization. FIFO can result in selling expired products, legal issues, or loss of trust.

https://doi.org/10.1016/j.jafr.2025.101848

Received 23 January 2025; Received in revised form 21 March 2025; Accepted 23 March 2025 Available online 24 March 2025 2666-1543/© 2025 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

^{*} Corresponding author. Division of Engineering Logistics, Lund University, Lund, Sweden. *E-mail address:* sandeep.jagtap@tlog.lth.se (S. Jagtap).

Implementing FIFO requires diligent tracking and management, which is complex and time consuming [5]. Also needs a lot of space to rotate inventory properly and it must be well-organized.

Adopting the FEFO approach in the PFSC management raises some practical problems that act as bottlenecks to implementing FEFO. As each method has its advantages and disadvantages, the FEFO approach, while boosting productivity by minimising wastage, is also difficult to implement due to various challenges discussed in the paper. A key challenge to implementing FEFO in PFSCs is the difficulty in constantly maintaining inventory and expiry date tracking [6]. Unlike items that don't go bad, perishables need close watching of their use-by dates at every step from production to retail [7]. This poses a significant drawback because different products have different shelf lives, and all stakeholders need reliable, relevant and up-to-date information about the product and its expiration date. Perishability of goods also depends upon the temperature at which they are stored and transported which adds cost and difficulty due to maintenance of cold chain throughout the process. Generally, these goods are stored and transported at temperatures of -20 °C to -24 °C [8]. Advanced technological solutions, such as automated inventory management systems and IoT-enabled sensor systems to track and measure the product and its perishability can be used to resolve the challenges which are outlined in the later sections of the paper [9]. However, these solutions have their own restrictions such as high implementation costs, data integration, and the need for employee training. These challenges act as a significant barrier to the widespread use of these technologies.

Fragmented coordination and collaboration among supply chain partners is also a prime reason that makes it difficult to apply FEFO technique in the PFSC [10]. This process involves several layers of the supply chain, including suppliers, manufacturers, distributors, and retailers, each with their own preferences and operational constraints [11]. Misaligned goals, lack of trust, and infrequent information flow can hinder collaboration and lead to suboptimal outcomes. Furthermore, non-standard regulatory compliances and practices across the supply chain network add another layer of complexity, as supply chain partners must account for differing requirements related to food safety, labelling, and inventory rotation.

Aligning the PFSC with sustainable development goals (SDGs) adds yet another dimension to the obstacles of FEFO implementation. Three important SDGs are covered in the study: SDG 12 (Responsible Consumption and Production), SDG 2 (Zero Hunger), and SDG 1 (No Poverty). The study finds the difficulties of implementing FEFO in PFSC. It seeks to resolve several issues. FEFO helps businesses save money by reducing food waste [12]. This supports SDG 1 by lowering costs and help fighting poverty. Essential food items become more available and cheaper [13]. FEFO helps with SDG 2 by sending out food items before they expire, reducing waste and getting more food to people. This helps fight hunger by making better use of available food [14]. This efficient channelling of food products helps meet demand consistently, alleviating food insecurity.

The study encourages responsible production, which advances SDG 12. It does this by selling and distributing items based on their expiration dates. Companies therefore improve their inventory turnover. They also cut down on the waste produced. It identifies the challenges and proposes solutions to it. This can be effective for making sustainable food supply chain methods. By implementing the findings, stakeholders in the perishable food industry may increase food security, reduce poverty, and advance sustainability. As sustainable practices become the norm and regulators increasingly demand sustainability in the supply chain, supply chain managers must balance the need for efficient inventory turnover with the minimization of food waste and environmental impact [15].

Due to the interconnectedness of the challenges and their impact on the implementation of FEFO in the PFSC, it is imperative to analyse and understand how each factor influences the others and which factor acts as a bottleneck to enacting FEFO in the supply chain. Through this paper, authors aim to analyse these challenges through a multi-layered approach after a comprehensive literature review including both research papers and industry insights. Therefore, by identifying the technological, collaborative, and sustainability issues in FEFO, this paper aims to develop a framework for addressing such bottlenecks and increasing the stability and efficiency of the PFSC. This study hence identifies two research questions to address the challenges faced in implementing FEFO:

RQ1. What are the key challenges in implementing FEFO for PFSCs? RQ2. How to analyse the influence of one factor on other factors?

To address RQ1 and RQ2, the paper also identifies the following research objectives:

RO1. To identify key challenges preventing implementation of FEFO in PFSC.

RO2. To develop network relationship map and level partitioning chart to analyse interrelationship between the identified challenges.

The study reviews major challenges including good inventory tracking, different product expiration dates, staff training and consumer related behaviour. The aim of the study is to analyse these challenges and how they influence each other as well as their impact on implementing FEFO in the PFSC.

This study proposes a unique method of inventory control and storage for the PFSC. Through FEFO, suppliers and retailers can maximize their profits by cutting down on costs and wastage. The study's scope also includes analysis of challenges preventing FEFO implementation. This paper leverages Decision Making Trial and Evaluation Laboratory (DEMATEL) and Total Interpretive Structural Modelling (TISM) and other Multi-criteria Decision Making (MCDM) techniques to rank the challenges based on their order of influence on other issues so that a broad view of which challenges can affect the other factors to the greatest degree can be obtained. This study adds to the existing literature by comprehensively analysing obstacles to integrating FEFO with the PFSC and identifying which of these challenges need to be solved first.

This paper follows the following structure: Section 1 elaborates on the introduction – giving a brief overview of the paper, the RQs identified and ROs the paper hopes to achieve. Section 2 focuses on the literature review for the study using the Preferred Reporting Items for Systematic reviews and Meta-Analyses (PRISMA) technique. Section 3 details the methodology used in the paper while Section 4 focuses on the data collection method along with the DEMATEL technique. Section 5 includes the results and discussion where matrix analysis is conducted; while Section 6 deals with the implications of the results along with the Level Partitioning Chart. Section 7 introduces future scope of the study and the conclusion while the final Section 8 includes the references.

2. Literature review

An extensive literature review using the PRISMA methodology, focusing FIFO, FEFO, DEMATEL, and TISM were sourced from various sources, with ten relevant papers selected for in-depth meta-data analysis. The PRISMA approach facilitated the systematic organization of selected papers to meet the objectives of this study is presented in Fig. 1. PRISMA is employed to ensure a systematic, transparent, and reproducible literature review process. Given the complexity of perishable food logistics, a structured approach is essential to comprehensively identify relevant studies on FEFO (First-Expired, First-Out) implementation challenges and PRISMA facilitates the elimination of selection bias, ensuring that only high-quality, peer-reviewed sources contribute to the analysis.

In recent years, there has been a surge in research interest surrounding PFSCs. This increase is evident from the growing body of



Fig. 1. PRISMA approach.

literature dedicated to optimizing inventory management strategies, such as FEFO and FIFO methods. Between 2022 and 2024, over 17,101 publications have focused on PFSC, reflecting its critical role in efficient supply chain management. FEFO-specific research has 191 publications focusing on its application in PFSC. It has shown a sign of growth in the field. 836 papers have addressed FIFO systems in PFSC. The integration of digital systems in FEFO is gaining momentum. There are 393 papers focused on using digital technologies to enhance FEFO in PFSC. 255 papers have been written sustainable development goals achieved in FEFO.

The application of the DEMATEL method has become a valuable tool in FEFO-related PFSC studies, as evidenced by 1090 publications. DEMATEL's ability to analyse complex causal relationships between factors influencing perishable inventory decisions has proven useful for researchers aiming to prioritize sustainability in PFSC [16]. This growing body of literature reflects the increasing interest in leveraging digital solutions and advanced decision-making frameworks to support FEFO in PFSC, addressing both efficiency and sustainability as key objectives in the perishable goods industry.

2.1. Perishable Food Supply Chain (PFSC)

A PFSC is a system for managing the journey of fresh food products. Perishable items like fruits, vegetables, dairy, and meat from production to the consumer are met through this system [17]. Unlike non-perishable products, these foods quickly lose quality therefore they should be handled carefully. To keep them fresh and safe the system takes decision about where to store the food, how to transport it and when to move it. Environmental conditions, like temperature, are crucial to food spoilage. High temperatures can quickly spoil the food [18].

There is rising significance of FEFO in controlling PFSCs. FEFO helps to improve inventory control [19]. It also helps in decreasing food waste as customer demands fresh goods [20]. Using advanced technological

innovations enhance shelf-life tracking and real time visibility of the product [21]. These technologies affect positively in streamlining FEFO procedures. Every year there is a steady rise in the quantity of research articles on PFSC management [22]. Integration of digital technology reduce food waste. Growing interest in sustainability have contributed to the increase in research papers [23]. Researchers are now focusing on how FEFO can maximize inventory turnover. They are studying how it minimizes food waste in interrupted supply networks. Supply chain management (SCM) has become digital because of this, supply chain relies heavily on digital solutions [24].

2.2. Challenges faced in PFSC

The PFSC faces number of challenges. The major factors that affect the SC are shelf life of the products and demand being unpredictable [25]. These problems lead to high food waste rates. Reliable and timely information is therefore important. This information helps improve resilience and sustainability in the PFSC [26]. Earlier, in the perishable food sector FIFO was used for managing inventory [27]. SCM techniques focused on optimizing transportation, inventory, and logistics. Earlier, systems lacked digital integration. They lacked the flexibility needed to handle large-scale disruptions [28]. Effective SCM can increase revenue. It does this by maximizing delivery schedules, decreasing waste. It raises the product market value [29]. Studies indicate that a well-structured PFSC can increase sustainability. The environmental impact increases by 108 % as a result of refrigeration. The sustainability of the SC is increased by 137 % with a 23 % increase in economic awareness. Costs, environmental effects, and social implications are all balanced by PFSC principles [30]. COVID-19 changed consumer behaviour in terms of perishable food [31]. Lockdowns made consumers choose durable goods. They preferred items like pasta, canned foods, and frozen products. As a result of restaurants closing, customers had to buy online and at supermarkets. They made fewer trips. Food waste increased

significantly due to hoarding and transportation interruptions. This happened because of supply chain constraints [32]. It caused perishable goods to be overbought or thrown out. These behaviours stressed PFSC. It influenced both demand and waste patterns [33]. Pricing for perishable food is difficult because these items are time sensitive. Products near their expiration dates their prices are reduced to boost sales and minimize waste [34]. This leads to different consumer behaviours. Price-sensitive shoppers are attracted to discounts which increases the demand [35]. Some focus on freshness and therefore avoid items that are close to expiration. Pricing changes, such as dynamic pricing, reflect supply chain disruptions [36]. They also indicate changes in consumer demand. It affects overall consumption patterns. This is true during events like the pandemic. Consumers make quicker decisions when products are perishable. Discounts or promotions entice them to buy, reducing losses for shops. The focus on lower prices can lead to impulsive or bulk buying. This may ultimately increase food waste. Buying too many discounted perishables might cause them to spoil before they are consumed. Sellers use dynamic pricing models based on product freshness, seasonality, and demand [36]. Issues in supply chain reflects in significant price changes. It happens due to delivery delays and changes in production costs. Panic buying during the pandemic impacts pricing [37]. Long-shelf-life items are preferred by consumers, which raises journey expenses. Food waste increases because of spoilage, leading to higher demand for fresh produce [38]. This causes shortages and drives up the prices of perishable goods. Table 1 lists the challenges identified that prevent FEFO implementation in PFSC.

2.3. Technologies supporting First Expired, First Out (FEFO)

Global supply chains are becoming more complex. Also, the demand for fresh and safe food products is on rise. Products having limited shelf life increases the risk in perishable food. Improper handling of these items results in financial loss, food waste and health risks to consumers.

The FEFO method helps reduce waste by selling older goods before newer ones. This method has its advantages but implementing FEFO has some challenges to tackle. Traditional supply chain systems often struggle to meet the simple needs of FEFO. Inventory control and handling of products with different expiration dates are some of them. Factors such as insufficient employee training and impulsive customer behaviour affect adopting FEFO. These challenges prompted us to explore new solutions. The solutions which create a bridge between theory and practice in FEFO implementation.

The rapid development of digital technologies is a unique way to overcome these barriers. Use of IOT and Blockchain technology management allows to increase transparency and efficiency. Using these technologies, supply chain managers can understand inventory levels. Understanding the product movements, allows better decision making.

2.4. Gaps in the literature

Research has been conducted on FEFO warehousing systems and PFSCs but there is limited literature on integrating FEFO in PFSC. Current literature does not include how advanced technologies like Blockchain and IoT can benefit and remove challenges from implementing FEFO in the PFSC. Another blind-spot is the behaviour of consumers and how it acts as a barrier to implementing FEFO in the supply chain.

3. Methodology

DEMATEL helps to analyse the interdependencies among factors in complex decision-making scenarios. Experts evaluate the direct influence of each criterion on others. A robust dataset is created which helps in producing a cause-effect diagram. This allows stakeholders to visualize the influence network. It also helps to identify core factors. By understanding the cause-effect relationships, an accurate assessment of correlation between factors can be obtained.

Table 1

Notation	Challenges Identified	Explanation	Source
21	High Costs in Implementing FEFO	Implementing the FEFO system incurs high costs due	[14,29, 39]
		to the need for specialized	
		inventory software and skilled workers, alongside	
		significant upfront expenses	
		for detailed expiration	
		tracking, posing a financial	
		challenge for small and medium-sized enterprises.	
22	Management of Suppliers	Effective FEFO	[12,23,
	for Effective FEFO	implementation requires	40]
	Implementation	thorough supplier	
		management and accurate information flow to ensure	
		timely deliveries and proper	
		expiration tracking.	
		Misalignment or poor	
		coordination with suppliers and distributors can hinder	
		traceability and disrupt the	
		FEFO process.	
23	Integration of FEFO in	Integrating FEFO into existing	[40-42]
	Existing Systems	inventory management systems and software can be	
		complex. FEFO requires	
		precise tracking of expiration	
		dates for all inventory items,	
		which can be complex and time-consuming, especially	
		for large inventories with	
		diverse products.	
24	Inventory Rotation	Constantly rotating inventory	[37,42,
		to ensure that the oldest products are sold first can be	43]
		challenging, especially for	
		businesses with limited	
25	Challenges in Effective	storage space. Effective FEFO storage	[12,44,
- 5	Storage Management for	management faces challenges	45]
	FEFO	like limited space, the need	
		for meticulous control over	
		conditions, and logistical difficulties in organizing	
		products by expiration dates.	
		These issues may require	
		significant redesigns of	
		storage systems, and inadequate solutions can lead	
		to waste and financial losses.	
26	Seasonal Availability and	Perishable goods often have	[14,23,
	Supply Variation	seasonal availability, making	46]
		it hard to maintain consistent stock levels. This variability	
		requires flexible and	
		responsive inventory	
27	Consumer Demand	management strategies. Sudden changes in consumer	[3,47,
/	Fluctuations	demand, influenced by	[3,47, 48]
		trends, seasons, or unforeseen	-
		events, can disrupt FEFO	
		management. For instance, a spike in demand for certain	
		spike in demand for certain seasonal fruits can lead to	
		stock imbalances and	
		difficulty in managing	
90	Concumer Behaviour	expiration dates.	٢4
28	Consumer Behaviour	Consumers often choose products from the back of the	[4, 49–51]
		shelf rather than the soon-to-	
		be expired front shelved	
		products due to wanting fresh products rather than old ones.	
		IN DOLLARS LAUPE LUAD OLD ODES	

Table 1 (continued)

Notation	Challenges Identified	Explanation	Source
Ρ9	Regulatory Compliance and Training	Ensuring compliance with regulatory requirements for food safety and quality control can be challenging. Staff must be adequately trained to understand and implement FEFO processes correctly. Ensuring consistent compliance across all levels of the organization is difficult and requires ongoing training and monitoring.	[52-54]
P10	Operational Efficiency	Managing inventory based on expiration dates can sometimes conflict with other operational efficiencies, such as minimising handling time or optimizing storage space. This can lead to increased labour costs and reduced overall efficiency	[19,55]
P11	Data Accuracy and Reliability	Implementing FEFO relies heavily on accurate and reliable data regarding the expiration dates of products. Inaccurate data can lead to expired products being shipped, resulting in customer dissatisfaction and potential health risks	[11,55, 56]
P12	System Maintenance	Maintaining the FEFO system and ensuring that it continues to function effectively over time can be challenging, especially if the system is complex or has multiple components.	[12,57]
P13	Technological Requirements for Inventory and Expiration Date Tracking	Proper tracking of inventory levels and expiration dates is crucial for FEFO implementation. Advanced technological solutions like warehouse management systems and inventory tracking software are often needed, requiring significant investment in hardware and software.	[14, 58–60]

3.1. Data collection

The required data regarding interconnectedness of factors was collected from industry experts and professionals. After filling out a questionnaire, the panel assigned a number from 0 to 5 based on the degree of influence of one factor on another. This rating was conducted for all 13 factors, leading to a 13×13 matrix from each expert whose details are mentioned in Table 2. To ensure fairness and no sampling bias, professionals from various sectors and industries were chosen based on random sampling. Table 2 displays the expert's details, including their field, years of experience and proficiency. The challenges listed in Table 1 were obtained by conducting a literature review of obstacles faced in implementation of FEFO method. The average matrix (B) is compiled which depicts pairwise comparison of influence of the factors based on scale scores assigned by the experts. For statistical significance, a panel of ten members was determined to be the most practicable sample size for this research. The purpose of this panel is to provide important information about the challenges faced while implementing FEFO as well as how these factors affect each other.

To analyse and overcome the difficulties in FEFO implementation, an understanding of the factors and their interdependencies is necessary so that identification of the primary challenges and their solutions can Table 2

Name	Organization	Designation	Experience (years)	Expertise
Expert 1	Dairy Industry	Head of Operations	13	Product rollout, Demand forecasting
Expert	Horticulture	Plantation	10	Crop production,
2	Industry	Manager		Staff management
Expert	FMCG	Category	19	Product
3	Industry	Manager		positioning, Competitive analysis
Expert	Supply Chain	Director of	22	Warehousing
4	Expert	Logistics		logistics, Transportation
Expert 5	Agricultural Industry	Manager	12	Production, logistics
Expert 6	Leather Industry	Plant Manager	8	Inventory control
Expert 7	Supermarket	Division Head	17	Product control, inventory rotation
Expert 8	Automotive Industry	Assembly Line Supervisor	10	Scheduling, Quality Control
Expert 9	Hotel	Manager	15	Logistics, management
Expert 10	Restaurant	Head Chef	18	Inventory, procurement, quality control

occur. Causal dependency is defined as the linkage between cause and effect of two factors. To analyse the causal dependency, many methods such as TISM, Structural Equation Modelling (SEM), and DEMATEL etc. can be utilized. To aid decision making and easily observe interdependency of factors, the use of DEMATEL technique which is an MCDM technique is utilized (Braga et al., 2021). It also allows us to reframe causes and consequences and the way they are arranged in the structural framework. In comparison to other MCDM techniques like Analytic Hierarchy Process (AHP) and Technique for Order Preference by Similarity to Ideal Solution (TOPSIS), DEMATEL offers clear benefits such as assisting in establishment of interdependence between factors and mapping the challenges in a way that demonstrates relative links between challenges in the form of a Network Relationship Map [61]. Apart from DEMATEL, a level partitioning chart is created in the study to divide the challenges P1 to P13 according to their criticality. This is done to provide a more robust and transparent road map for addressing the obstacles which is a component of the TISM approach. Fig. 2 explains the step by step approach for the research methodology.

Step-1: An expert panel is organized to weigh in on the correlation of factor i on other all other factors j, denoted by a_{ij} . After the data is being collected the panel will rate the challenges on a scale of 0–5 as follows: 0 – no relation, 1 – minimal relation, 2 – low relation, 3 – moderate relation, 4 – high relation and 5 – maximal relation. As each expert was asked to rate the relationship between factors, a matrix with values from the given scale $X^k = [x_{ij}^k]$ was created. Using Equation (1), an average matrix B was created as an aggregate of the expert matrices.

$$\mathbf{B} = \begin{bmatrix} \mathbf{a}_{ij} \end{bmatrix} = \frac{1}{n} \sum_{i=1}^{n} \mathbf{x}_{ij}^{k} \tag{1}$$

Step-2: The average B matrix undergoes normalization to adjust values in a range of 0–1 using Equations (2) and (3).

$$N = \frac{B}{u}$$
(2)





$$u = 1 \le i \le n \left\{ \sum_{j=1}^{n} a_{ij} \right\}$$
(3)

Step-3: Using the Normalised matrix formed, the Total Relationship Matrix (TRM) is calculated via Equation (4) where 'I' represents the identity matrix. Overall relationship between each set of factors taken under consideration is represented by the TRM.

$$\mathbf{T} = \mathbf{N} \left(\mathbf{I} - \mathbf{N} \right)^{-1} \tag{4}$$

Step-4: To form the matrices, the sums of rows and columns are calculated using Equations (5) and (6). Factors such as dependency and driving power are represented by the matrices.

$$K = \left[\sum_{j=1}^{n} m_{ij}\right] = (k_1, k_2, \dots, k_i, \dots, k_n)$$
(5)

$$J = \left[\sum_{i=1}^{n} m_{ij}\right]' = (j_1, j_2, \dots, j_i, \dots, j_n)$$
(6)

$$M = m_{ij}$$
 $(i, j) = 1, 2, 3, ..., n$ (7)

Step-5: Threshold value (α) is calculated using Equation (7) for DEMATEL technique.

$$= \frac{\sum_{j=1}^{n} \sum_{i=1}^{n} k_{ij}}{n^2}$$
(8)

Step-6: Using K and J, cause and effect relationship is obtained by plotting the set

 $\label{eq:risk} \begin{array}{l} \{r_i+s_i,\,r_i-s_i\},\,\forall i \ \epsilon \ n. \ X-Axis=(r_i+s_i),\,Y-Axis=(r_i-s_i). \end{array}$ Digraph is a valuable tool for decision making as it clearly defines the interrelationship between the challenges.

3.1.1. Case study of dairy industries implementing FEFO & FIFO

In the dairy supply chain, effective inventory management is critical due to the perishable nature of many products, such as milk, cream, and yogurt. FIFO is the most widely used inventory management system to ensure product quality. Opting for a different inventory control system like FEFO comes with many benefits such as reduced waste, lower cost and optimized supply chain efficiency. FIFO is particularly important for dairy products with longer shelf life, such as cheese or yogurt. In this system, the oldest products in inventory are sold or used first, ensuring that products reach consumers before they lose freshness. FIFO is essential in maintaining consistent product quality and is particularly useful for dairy items that have a relatively longer shelf life compared to highly perishable products like milk. On the other hand, FEFO is applied in managing products that are highly perishable, such as milk and fresh cream, which have a short shelf life. In this system, products that are closest to their expiration date are prioritized for distribution, ensuring that older stock is consumed or sold first. This method is particularly beneficial in dairy operations where products need to be consumed quickly to avoid spoilage and minimize waste. FEFO allows dairy suppliers to align product distribution with expiration dates, significantly reducing the risk of expired goods reaching consumers. By effectively tracking and managing expiration dates, FEFO helps dairy companies improve their sustainability by cutting down on food waste and enhancing the overall efficiency of the supply chain.

4. Results and discussion

4.1. Study of correlation between factors

DEMATEL technique is facilitated by a panel of industry experts who provide data on interdependence of factors. The expert's data are listed in Tables A3-A7 in the Appendix. The rating criteria and scale listed depicts the degree of influence of one factor on another and allows us to use DEMATEL technique and Level Partitioning Chart to identify critical factors.

Step 1: Collecting data from experts.

Equation (1) is used on the gathered data, to create the average B

α

The values provided by experts are listed in Table 3.

Step 2: Normalising average matrix

Add all the row values to obtain row-wise sum and divide the average matrix B by the highest row sum. The new matrix obtained, N, is the normalised matrix. Matrix N is displayed in Table 4.

Step 3: TRM Computation

Using Eq. (4), the normalised matrix N is converted into the total relation matrix T. The T Matrix is computed using MATLAB to evaluate the normalised matrix and calculate cell values using equation. T Matrix is shown in Table 5.

Step 4: Calculation of row sum and column sum. (Sr and Sc). The sum of rows and sum of columns is calculated for the Total Relation Matrix (T). Sr and Sc values are calculated and Table 6 displays these values next to the matrix T.

Step 5: Determine the alpha threshold value

As we have 13 key factors, they form a 13×13 matrix and we use n = 169 while computing the threshold value, alpha. This value is found to be 0.368462. The alpha threshold is determined by taking the average of all matrix values as it determines a base level of interdependency below which correlation is not significant. The alpha values are fitted into the T Matrix and all values below threshold are eliminated, thus making the Adjusted Total Relation Matrix. The Adjusted TRM is shown in Table 7.

Step 6: Building a directed causal network

The values of Sr + Sc and Sr–Sc are used as the horizontal and vertical axes, respectively, in the causal plot for the DEMATEL process. Consequently, y = Sr - Sc and x = Sr + Sc. Table 7 presents the significant relationship matrix S, derived after applying the threshold value used to construct the directed causal graph. The interrelationships between the obstacles to implementation of the FEFO method in the PFSC are better understood with the aid of this map. Studying the essential barrier among the 13 problems described is aided by the plot as well. The digraph is displayed in Fig. 3, and the final causal matrix is displayed in Table A2.

Adopting advanced inventory management techniques like the FEFO system has become essential as the perishable food industry faces increasing pressure to increase operational efficiency while reducing waste. By shipping goods as close to their expiration date as possible, this strategy lowers the possibility of health risks and minimizes food waste. Traditional inventory techniques like FIFO (First In First Out) and LIFO (Last In First Out) do not take product perishability into account as much as FEFO does. Real-time expiration date monitoring, the

Journal of Agriculture and Food Research 21 (2025) 10	01848
---	-------

requirement for technology advancements, and efficient stakeholder collaboration throughout the supply chain are some of the challenges that FEFO implementation faces.

Because perishable products have short shelf lives and need continual supervision, the PFSC is complicated and depends on precise data for decision-making. Although there are drawbacks, including high implementation costs and the requirement for expert labour, integrating IoT-enabled sensors with automated inventory systems can offer a solution. Additionally, there is frequently a lack of coordination between SC participants, such as distributors, retailers, and suppliers, which results in inefficiencies during the FEFO implementation process.

Using the DEMATEL method, which aids in visualising and quantifying the cause-and-effect links among the different obstacles, this study seeks to identify and assess these issues. The study will produce a roadmap for removing the obstacles to FEFO implementation by gathering data from industry experts and conducting pairwise comparisons, which will ultimately result in a PFSC that is more robust and sustainable.

4.2. Level partitioning on PFSC

For FEFO to be used, inventory levels and expiration dates must be monitored precisely. Often, sophisticated technological solutions are needed, including warehouse management systems and inventory tracking software such as Katana and NetSuite. Significant investments are required for both software and hardware. A comprehensive examination of the difficulties encountered while applying the FEFO strategy in the PFSC may thus be conducted by combining the DEMATEL and TISM approaches. These tools facilitate the examination, visualisation and analysis of key challenges to FEFO implementation in the PFSC. Given the short shelf life of products, specific obstacles must be addressed to ensure effective inventory management using methods like FEFO. It is crucial to identify which barriers has the most impact and are more influenced by external factors to successfully implement FEFO. By solving most important issues that indirectly affect other, less critical factors, FEFO can be smoothly implemented with the least amount of effort and greatest efficiency.

To tackle these challenges, an average matrix B is drawn based on insights obtained from a panel of industry experts. This matrix is then normalised to create a Total Relation Matrix (TRM), which illustrates the cause-and-effect relationships, based on an approach used in prominent research articles. These challenges are then categorised into levels where their rank priority is decided using the Total Relationship Matrix based on DEMATEL methodology. Factors related to FEFO implementation, such as inconsistent stakeholder coordination, technological constraints, and inventory management issues etc. are ranked according to their significance and influence on other challenges.

Issues like P13 may rank highly because they affect several other problems that are ranked lower. It's important to solve operational and technological problems first. This ensures smooth FEFO functioning

Table 3	
Average B matrix with interdependencies ranked.	

	P1	P2	P3	P4	P5	P6	P7	P8	Р9	P10	P11	P12	P13
P1	1	2	2	1	4	0	2	4	2	2	3	2	1
P2	0	1	2	0	4	2	3	4	0	3	2	3	2
P3	4	3	1	4	0	1	3	4	1	0	3	4	2
P4	3	1	3	1	1	3	4	2	1	4	1	4	3
P5	3	3	4	2	1	4	0	1	3	3	3	3	2
P6	3	3	0	1	1	1	1	2	1	0	1	3	4
P7	3	4	3	4	4	1	1	2	0	1	2	0	1
P8	0	1	0	4	1	1	1	1	2	1	0	2	4
P9	3	1	1	1	3	1	4	2	1	0	3	4	3
P10	0	2	3	1	2	1	4	0	0	1	4	4	2
P11	4	2	3	0	3	1	3	0	1	3	1	1	3
P12	3	1	3	4	0	1	1	3	4	0	3	1	4
P13	4	2	0	0	4	2	1	3	4	4	3	1	1

Table 4

Normalised matrix (N).

	P1	P2	Р3	P4	P5	P6	P7	P8	Р9	P10	P11	P12	P13
P1	0.03125	0.0625	0.0625	0.03125	0.125	0	0.0625	0.125	0.0625	0.0625	0.09375	0.0625	0.03125
P2	0	0.03125	0.0625	0	0.125	0.0625	0.09375	0.125	0	0.09375	0.0625	0.09375	0.0625
P3	0.125	0.09375	0.03125	0.125	0	0.03125	0.09375	0.125	0.03125	0	0.09375	0.125	0.0625
P4	0.09375	0.03125	0.09375	0.03125	0.03125	0.09375	0.125	0.0625	0.03125	0.125	0.03125	0.125	0.09375
P5	0.09375	0.09375	0.125	0.0625	0.03125	0.125	0	0.03125	0.09375	0.09375	0.09375	0.09375	0.0625
P6	0.09375	0.09375	0	0.03125	0.03125	0.03125	0.03125	0.0625	0.03125	0	0.03125	0.09375	0.125
P7	0.09375	0.125	0.09375	0.125	0.125	0.03125	0.03125	0.0625	0	0.03125	0.0625	0	0.03125
P8	0	0.03125	0	0.125	0.03125	0.03125	0.03125	0.03125	0.0625	0.03125	0	0.0625	0.125
P9	0.09375	0.03125	0.03125	0.03125	0.09375	0.03125	0.125	0.0625	0.03125	0	0.09375	0.125	0.09375
P10	0	0.0625	0.09375	0.03125	0.0625	0.03125	0.125	0	0	0.03125	0.125	0.125	0.0625
P11	0.125	0.0625	0.09375	0	0.09375	0.03125	0.09375	0	0.03125	0.09375	0.03125	0.03125	0.09375
P12	0.09375	0.03125	0.09375	0.125	0	0.03125	0.03125	0.09375	0.125	0	0.09375	0.03125	0.125
P13	0.125	0.0625	0	0	0.125	0.0625	0.03125	0.09375	0.125	0.125	0.09375	0.03125	0.03125

Table 5

Total relation matrix calculated.

	P1	P2	Р3	P4	Р5	P6	P7	P8	P9	P10	P11	P12	P13
P1	0.3803	0.3545	0.355	0.3024	0.4432	0.2187	0.3653	0.4378	0.3048	0.3221	0.4222	0.4075	0.3884
P2	0.3452	0.3244	0.3469	0.272	0.4331	0.2771	0.3823	0.4316	0.2423	0.3447	0.3863	0.4285	0.4154
P3	0.5209	0.4199	0.358	0.431	0.3714	0.272	0.4408	0.497	0.3091	0.3012	0.4594	0.5081	0.47
P4	0.5085	0.3822	0.4313	0.3532	0.4111	0.34	0.4815	0.445	0.3174	0.4212	0.4265	0.527	0.5103
P5	0.5271	0.45	0.4713	0.3801	0.4231	0.3819	0.3823	0.431	0.3879	0.4039	0.5016	0.5244	0.5021
P6	0.3729	0.3233	0.2252	0.2379	0.3001	0.2037	0.2694	0.329	0.2373	0.2159	0.2972	0.3631	0.4088
P7	0.4484	0.4256	0.3944	0.3923	0.4526	0.2603	0.3463	0.3958	0.2422	0.3097	0.3971	0.3628	0.3913
P8	0.2574	0.2354	0.2024	0.3079	0.2628	0.1915	0.2512	0.2614	0.2428	0.2253	0.2369	0.3109	0.3799
P9	0.4714	0.3429	0.3386	0.3151	0.4383	0.2572	0.434	0.4021	0.2956	0.2762	0.4411	0.4732	0.4639
P10	0.3481	0.348	0.3768	0.2896	0.3684	0.2361	0.4117	0.3068	0.2295	0.2785	0.4402	0.442	0.3975
P11	0.4699	0.3597	0.3815	0.2591	0.4204	0.2415	0.3916	0.322	0.2675	0.3501	0.3684	0.3703	0.4292
P12	0.4828	0.3422	0.3938	0.4063	0.3553	0.2607	0.3699	0.4462	0.3901	0.2864	0.4468	0.4063	0.5094
P13	0.4986	0.3816	0.3178	0.2796	0.4844	0.2954	0.3652	0.433	0.3869	0.4054	0.4596	0.4124	0.4188

Tabl	le 6
------	------

TRM with row sum and column sum.

	P1	P2	Р3	P4	Р5	P6	P7	P8	Р9	P10	P11	P12	P13	ROW_SUM
P1	0.3803	0.3545	0.355	0.3024	0.4432	0.2187	0.3653	0.4378	0.3048	0.3221	0.4222	0.4075	0.3884	4.7022
P2	0.3452	0.3244	0.3469	0.272	0.4331	0.2771	0.3823	0.4316	0.2423	0.3447	0.3863	0.4285	0.4154	4.6298
P3	0.5209	0.4199	0.358	0.431	0.3714	0.272	0.4408	0.497	0.3091	0.3012	0.4594	0.5081	0.47	5.3588
P4	0.5085	0.3822	0.4313	0.3532	0.4111	0.34	0.4815	0.445	0.3174	0.4212	0.4265	0.527	0.5103	5.5552
P5	0.5271	0.45	0.4713	0.3801	0.4231	0.3819	0.3823	0.431	0.3879	0.4039	0.5016	0.5244	0.5021	5.7667
P6	0.3729	0.3233	0.2252	0.2379	0.3001	0.2037	0.2694	0.329	0.2373	0.2159	0.2972	0.3631	0.4088	3.7838
P7	0.4484	0.4256	0.3944	0.3923	0.4526	0.2603	0.3463	0.3958	0.2422	0.3097	0.3971	0.3628	0.3913	4.8188
P8	0.2574	0.2354	0.2024	0.3079	0.2628	0.1915	0.2512	0.2614	0.2428	0.2253	0.2369	0.3109	0.3799	3.3658
P9	0.4714	0.3429	0.3386	0.3151	0.4383	0.2572	0.434	0.4021	0.2956	0.2762	0.4411	0.4732	0.4639	4.9496
P10	0.3481	0.348	0.3768	0.2896	0.3684	0.2361	0.4117	0.3068	0.2295	0.2785	0.4402	0.442	0.3975	4.4732
P11	0.4699	0.3597	0.3815	0.2591	0.4204	0.2415	0.3916	0.322	0.2675	0.3501	0.3684	0.3703	0.4292	4.6312
P12	0.4828	0.3422	0.3938	0.4063	0.3553	0.2607	0.3699	0.4462	0.3901	0.2864	0.4468	0.4063	0.5094	5.0962
P13	0.4986	0.3816	0.3178	0.2796	0.4844	0.2954	0.3652	0.433	0.3869	0.4054	0.4596	0.4124	0.4188	5.1387
COL_SUM	5.6315	4.6897	4.593	4.2265	5.1642	3.4361	4.8915	5.1387	3.8534	4.1406	5.2833	5.5365	5.685	

Table 7

Adjusted total relation matrix as per alpha threshold.

	P1	P2	Р3	P4	Р5	P6	P7	P8	Р9	P10	P11	P12	P13
P1	-				0.4432			0.4378			0.4222	0.4075	0.3884
P2		-			0.4331		0.3823	0.4316			0.3863	0.4285	0.4154
P3	0.5209	0.4199	-	0.431	0.3714		0.4408	0.497			0.4594	0.5081	0.47
P4	0.5085	0.3822	0.4313	-	0.4111		0.4815	0.445		0.4212	0.4265	0.527	0.5103
P5	0.5271	0.45	0.4713	0.3801	-	0.3819	0.3823	0.431	0.3879	0.4039	0.5016	0.5244	0.5021
P6	0.3729					-							0.4088
P7	0.4484	0.4256	0.3944	0.3923	0.4526		-	0.3958			0.3971		0.3913
P8								-					0.3799
P9	0.4714				0.4383		0.434	0.4021	-		0.4411	0.4732	0.4639
P10			0.3768		0.3684		0.4117			-	0.4402	0.442	0.3975
P11	0.4699		0.3815		0.4204		0.3916				-	0.3703	0.4292
P12	0.4828		0.3938	0.4063			0.3699	0.4462	0.3901		0.4468	-	0.5094
P13	0.4986	0.3816			0.4844			0.433	0.3869	0.4054	0.4596	0.4124	-



Fig. 3. Network relation map.

before addressing lower-level issues like P10. Supply chain managers can then create a roadmap that concentrates on resolving the most important obstacles first by breaking these issues down into tiers, which will make the FEFO method implementation easier. This is done via the Level Partitioning Chart made using the TRM. Table 8 shows the level partitioning matrix which describes how the challenges are being influenced and are getting influences by other challenges.

The challenges, after performing a level partitioning procedure, are separated into seven levels. At top there is Consumer Behaviour (P8), Data Accuracy and Reliability (P11) and Technological Requirements for Inventory and Expiration Date Tracking (P13). These challenges at the top-level show that they are the least influential on other challenges,

Table	8				
Level	partitioning	matrix	with	final	iteration.

- . .

Challenges	Influencing	Influenced By	Intersection Set	Levels
P1	P1, P5, P8, P11, P12, P13	P1, P3, P4, P5, P6, P7, P9, P11, P12, P13	P1, P5, P11, P12, P13	2
P2	P2, P5, P7, P8, P11, P12, P13	P2, P3, P4, P5, P7, P13	P2, P5, P7, P13	5
P3	P1, P2, P3, P4, P5, P7, P8, P11, P12, P13	P3, P4, P5, P7, P10, P11, P12	P3, P4, P5, P7, P11, P12	6
P4	P1, P2, P3, P4, P5, P7, P8, P10, P11, P12, P13	P3, P4, P5, P7, P12	P3, P4, P5, P7, P12	6
P5	P1, P2, P3, P4, P5, P6, P7, P8, P9, P10, P11, P12, P13	P1, P2, P3, P4, P5, P7, P9, P10, P11, P13	P1, P2, P3, P4, P5, P7, P9, P10, P11, P13	5
P6	P1, P6, P13	P5, P6	P6	3
P7	P1, P2, P3, P4, P5, P7, P8, P11, P13	P2, P3, P4, P5, P7, P9, P10, P11, P12	P2, P3, P4, P5, P7, P11	3
P8	P8, P13	P1, P2, P3, P4, P5, P7, P8, P9, P12, P13	P8, P13	1
Р9	P1, P5, P7, P8, P9, P11, P12, P13	P5, P9, P12, P13	P5, P9, P12, P13	4
P10	P3, P5, P7, P10, P11, P12, P13	P4, P5, P10, P13	P5, P10, P13	7
P11	P1, P3, P5, P7, P11, P12, P13	P1, P2, P3, P4, P5, P7, P9, P10, P11, P12, P13	P1, P3, P5, P7, P11, P12, P13	1
P12	P1, P3, P4, P7, P8, P9, P11, P12, P13	P1, P2, P3, P4, P5, P9, P10, P11, P12, P13	P1, P3, P4, P9, P11, P12, P13	4
P13	P1, P2, P5, P8, P9, P10, P11, P12, P13	P1, P2, P3, P4, P5, P6, P7, P8, P9, P10, P11, P12, P13	P1, P2, P5, P8, P9, P10, P11, P12, P13	1

but they are influenced the most by other challenges (Fig. 4). Table A1 explains the interrelationship between the barriers after level partitioning. The lack of access to technology for real-time, efficient monitoring of inventory and expiry date tracking is one of the key challenges holding back implementation of FEFO in PFSCs.

This, combined with the challenge of consumer behaviour and data accuracy illustrates that P8, P11 and P13 are not the crucial pivotal challenges, but they are the challenges influenced by the highest number of factors. P13 is influenced by every other 12 challenges identified in the model, inferring that finding solutions for the other challenges would help solve the challenges posed by barrier P13. In Level 1 challenges, consumer behaviour comes next (P8). P8 is influenced by every other challenge except P6. Similarly, P11 is influenced by every factor apart from P6 and P8 - implying that solving these challenges will automatically resolve the barriers at the topmost level of the level partitioning chart.

At the second level, high cost of implementing FEFO (P1) is the only challenge, and it highlights how cost influences various factors such as P8, P9, P11, P12, P13 etc. while also being influenced by factors such as P3, P4, P5, P6 etc. To solve issues on this layer, cost optimization needs to be carried out by resolving the factors that influence cost - hence offering partial solutions to factors being influenced by cost such as P13.

The third level includes challenges such as seasonal availability and supply variation (P6) and consumer demand fluctuations (P7). Level 3 challenges are based on the basic stumbling block of demand and supply variability. As the perishable food industry is a dynamic and timesensitive market, demand and supply inevitably fluctuates - with certain products only available in certain seasons and demand also depending on time periods and conditions. So, these challenges are influenced by other factors like inventory rotation (P4), storage management (P5) etc. Solving these challenges by accounting for seasonal variability and shifts in supply and demand can also influence challenges in the previous layers.

The fourth level comprises roadblocks like regulatory compliance and training (P9) and system maintenance (P12). This layer addresses the issue of permanence and maintenance necessary for implementing FEFO. As inventory control in FEFO is much more complex due to integration of time dependent processes and expiration date tracking, maintaining these systems for a long duration also poses a substantial challenge and as new employees are not familiar with the workflow and FEFO method, once experienced employees leave or are no longer available, the system can crumble and decay over time. To solve this issue, regular training and development of standardised workflows is necessary along with periodic maintenance and checking. The challenges on this layer are influenced by factors like P2, P11 and P13.

The challenges in the fifth layer are Management of Suppliers for Effective FEFO Implementation (P2) and Challenges in Effective Storage Management for FEFO (P5). These challenges are related to the issues faced in managing the supply chain and complexity of inventory control, expiry date tracking and maintaining clear communication lines across all layers of the SC. These are some of the fundamental challenges that need to be solved before FEFO can be implemented with optimal efficiency. They are also influenced by other challenges such as P3, P4, P5, P7, P9, P10, P11 and P13 showing their complex interdependence on other factors. Integration of real time data monitoring systems, IoT and Blockchain are possible solutions for the challenges in this layer.

The sixth layer is made up of challenges such as Integration of FEFO in existing systems (P3) and inventory rotation (P4). These challenges illustrate the difficulties faced while shifting from one inventory management system to another as well as issues faced in inventory rotation in the FEFO system, especially with limited storage space. A major shift in company policy, working methodology, standards etc. needs to take place if FEFO is to be successfully implemented. This can be extremely challenging for businesses with established methods and traditions and requires a lot of time and resource investment. Possible solutions might be to shift focus to FEFO on a project by project basis while adopting it as



Fig. 4. Level partitioning chart.

the standard model of inventory control for future products or shipments. These challenges are influenced by challenges like P5, P7 and P12. They also influence a majority of challenges such as P1, P2, P5 P7, P8, P10, P11, P12 and P13 thus showcasing their importance and how their resolution can help solve the other challenges.

The seventh and final layer has only one challenge i.e. operational efficiency (P10). Managing inventory based on expiration dates can sometimes conflict with other operational efficiencies, such as minimising handling time or optimizing storage space. This can lead to increased labour costs and reduced overall efficiency. This is a key challenge as it directly conflicts with the core mandate of the FEFO method and hence is influenced by very few factors like P4, P5 and P13. On the other hand, it influences other factors like P4, P5, P7, P11, P12 and P13. Improving operational efficiency while sticking to the core concept of FEFO by eliminating waste and resource allocation to other non-critical areas is one of the possible ways in which this challenge can be resolved. The influence of factors and their levels of interdependence are shown in Fig. 4.

5. Implications

5.1. Theoretical implications

The FEFO method is gaining traction in the PFSC industry because of its potential for food safety and minimising waste. This study offers insight into the challenges and the way they are interconnected during the implementation of FEFO in a dynamic and time-sensitive industry like perishable foods. Through this research, the observation and analysis of complex roadblocks to integrating FEFO in the supply chain is done and a framework for implementing time sensitive processes in the SC is laid out.

This paper identifies and prioritises critical challenges such as

technological readiness, accuracy of expiration tracking systems, and the need for seamless communication across the SC. Building an efficient FEFO system that enhances product quality, reduces wastage and increases customer satisfaction is highly dependent on factors such as seamless communication, technological readiness and accurate inventory control. Moreover, the research highlights the importance of transparency and open communications at all layers of supply chain and real-time data sharing, which allows easy integration of real time data monitoring systems in applications such as inventory tracking and control.

This research also offers a base from which future studies into supply chain management can be carried out by using MCDM techniques like TISM and DEMATEL. By focusing on real time monitoring and data integration, this research increases the understanding on topics like inventory management and the logistics of time-sensitive products. It introduces an improved understanding of the interaction between information flow, product life cycle, and decision-making processes in inventory management. It also delves into how 'soft' problems like consumer behaviour, shelf stocking etc. can impact the implementation of FEFO.

The findings also highlight the role of communication and cooperation between supply chain partners in reducing inefficiencies and preventing bullwhip effect. FEFO allows fast moving flow of goods and information and also increases the efficiency of the PFSC, hence acting as a tool for optimizing the supply chain operations.

5.2. Practical implications

The FEFO approach has a direct impact on operating efficiency, waste reduction, and product quality, all of which present substantial hurdles to implementation of FEFO. However, this strategy provides enormous benefits to both enterprises and consumers. The core directive

J. Kandasamy et al.

of FEFO is that near-expiry products be sold or consumed first, preventing the delivery of expired stock and useless inventory, resulting in less waste. This is consistent with the growing need for sustainability in both commercial and individual contexts, as defined by the Sustainable Development Goals (SDGs). Waste reduction and improved product quality are the core benefits offered by the FEFO approach.

FEFO implementation requires real-time tracking systems capable of monitoring product expiration dates and inventory levels. Using IoT, barcodes, RFID tagging etc. are some of the main ways with which ensuring perishable items are accurately tracked from the moment they enter the warehouse until they reach the consumer are done. Integrating these technologies into the FEFO system also comes with logistical challenges such as revamping current inventory control methods, high technological requirements, complex inventory tracking etc. but offers significant benefits in the form of improved inventory turnover, product quality, efficiency, reduced wastage etc. This paper lays the framework from where future research on solutions can be conducted for adopting FEFO in the PFSC.

Just like any inventory and supply chain control system, FEFO relies on smooth flow of materials and information across all strata of the supply chain. Collaboration and transparency are some of the key pillars on which a well implemented and efficient FEFO system is built. Clear communication regarding product expiry date, shelf life and stock levels allows movement of inventory in an efficient manner through the supply chain, minimising the risk of spoilage. Other challenges such as reconfiguring warehouse and retail operations to shift from current methodology (FIFO) to proposed methodology (FEFO) is necessary. This might mean changing storage conditions, conventions, stock rotation, staff training methods as well as supply chain management.

Implementing FEFO not only benefits consumers and people due to its sustainability but also helps business and companies as it allows reduced waste-related and spoilt inventory costs and boosts consumer satisfaction. Consumers get the satisfaction of getting fresher products, while companies gain a competitive advantage by minimising waste, increasing profits in the long term and enhancing their reputation and brand identity by conforming to health and safety standards while simultaneously boosting the sustainability of their operations. Moreover, FEFO supports sustainability goals SDG 9, 11 and 12 by directly addressing food waste, conforming business practices with the initiative to promote sustainable consumption and production.

By adopting the FEFO method, businesses can optimize their supply chains, ensuring that perishable goods are available as needed and are in optimal condition with minimal spoilage. This makes FEFO a more consumer-centric supply chain model that benefits both the business and the consumer.

This article focuses on the implementation of the FEFO approach in PFSC, outlining and discussing 13 significant difficulties based on a thorough literature assessment and input from industry specialists. FEFO is a strategy that offers significant advantages over traditional approaches, such as FIFO and LIFO, by taking expiration dates into account while dispatching; it eliminates health concerns and food waste. However, FEFO faces substantial implementation obstacles, including technological limits such as the requirement for enhanced tracking systems as well as issues with efficient collaboration among many parties. To solve these hurdles, the research used the DEMATEL technique, which results in a relational diagram that depicts the causal linkages between the obstacles.

According to the report, the most major problem is proper storage management, which is driven by a variety of interconnected elements such as inventory tracking technology requirements. By categorizing these barriers hierarchically, this study provides stakeholders with a methodical framework for directing efforts ranging from critical technological investments to synergistic improvements throughout the SC. These findings highlight the importance of balancing FEFO implementation tactics with technology requirements, interdisciplinary collaboration, and sustainability. Building off of this research, a more thorough analysis would include engaging a greater, more diverse panel of industry experts, optimizing FEFO to include identification of industry specific changes, and an alternative approach that has manageable barriers or greater efficiency. This thesis offers PFSC managers a systematic guide that allows them to adopt the FEFO method to increase sustainability and efficiency while reducing waste.

6. Conclusions

The FIFO method is usually used for inventory rotation and control in PFSCs but it is insufficient for management of perishable goods because of its challenges such as imprecise tracking of spoilt goods and wastage. This not only increases spoilage but also increases the hazards associated with consumer health and regulatory compliance. Cases such as spoilage, food poisoning, health risks etc. are a cause for concern. Relying on FIFO, businesses face challenges such as loss of inventory, inefficiency and higher overhead costs thus impacting profitability and undermining supply chain sustainability. These limitations can be addressed by shifting to a new inventory management system such as FEFO which promises better results for PFSCs. This study identified two main research objectives which are identification and analysis of key challenges blockading implementation of FEFO and analysing interdependencies between these factors. This study adhered to strict methodology and proper scientific method. The literature review was conducted according to the PRISMA technique, and a panel of experts were consulted before the challenges could be rated. MCDM techniques such as DEMATEL and TISM were used to analyse and rate the 13 key challenges identified that prevent implementation of FEFO in PFSC. Causal dependencies between the factors were identified and influence of one factor on another was documented. DEMATEL was used to facilitate the identification of the challenges while TISM was used to form the causal diagram from which Network Relationship Map and Level Partitioning Chart were created that visually depicted correlation between challenges and level of influence. This structured approach allowed for the identification of key challenges such as effective storage management, high cost of implementing FEFO, consumer behaviour, technological dependence etc. The results identified that the most critical challenge was Effective Storage Management and tracking of expiry dates with an alpha threshold value of 0.368. This finding aligns with the FEFO methodology as proper storage and expiration date tracking is the core component for FEFO implementation. The findings from this study successfully aligned with the research objectives as a fair and accurate assessment of the key challenges was conducted and their interdependencies were calculated. This study builds on previous research conducted on general inventory strategies for perishable food items. Unlike current literature which focused on the FIFO approach or proposed theoretical frameworks, this research delves into the specifics of the FEFO method and proactively aims to identify challenges faced while implementing FEFO. Practical considerations such as consumer behaviour and difficulties in expiration date tracking were also considered and correlation between factors was identified allowing for smoother implementation of FEFO in PFSC in the future. The implications of this research extend beyond PFSCs and are relevant to a variety of industries that require stringent inventory management. The study's insights are particularly useful for pharmaceuticals, healthcare, and retail sectors where product freshness and regulatory compliance are critical. Additionally, logistics providers, cold storage facilities, and distributors can benefit from adopting the FEFO framework to minimize waste, increase product quality, and improve overall supply chain efficiency. By focusing on core elements such as data accuracy and storage optimization, this study can serve as a blueprint for other industries managing perishable inventory, thereby contributing to broader sustainability and waste reduction goals as well as target SDG 1 (No Poverty), SDG 2 (Zero Hunger) and SDG 12 (Responsible Consumption and Production). While the study presents valuable insights, it has some limitations. This study is primarily theoretical, with no practical

implementation or pilot testing of the proposed FEFO roadmap. Consequently, while the recommendations are grounded in thorough analysis, real-world testing is required to validate the proposed model's effectiveness. Furthermore, the study does not extensively explore emerging technologies like IoT, AI, or blockchain, which could provide dynamic solutions to challenges in data accuracy and inventory tracking. Future research should focus on testing the proposed roadmap in actual supply chain settings to assess its practicality and effectiveness. Pilot programs in sectors such as food retail or pharmaceuticals could yield valuable insights and highlight additional challenges or considerations. Future studies might also explore the scalability of FEFO practices for large organizations, as well as the potential integration of automated sorting systems to streamline inventory management in highly complex supply chains.

CRediT authorship contribution statement

Jayakrishna Kandasamy: Writing - original draft, Visualization,

Journal of Agriculture and Food Research 21 (2025) 101848

Validation, Methodology, Investigation, Formal analysis, Data curation, Conceptualization. **K.E.K. Vimal:** Writing – original draft, Visualization, Validation, Investigation, Formal analysis, Data curation. **Aditya Pratap Singh:** Writing – original draft, Formal analysis, Data curation. **Aaryan Magnani:** Writing – original draft, Investigation, Formal analysis, Data curation. **Ameya Gokhale:** Writing – original draft, Investigation, Formal analysis, Data curation. **Sandeep Jagtap:** Writing – review & editing, Supervision, Project administration, Funding acquisition, Formal analysis.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Annexure.

 Table A1

 Inference from level partitioning chart for each causal linkage

9 High costs in implementing FEPO affect data accuracy and reliability. 4 High costs in implementing FEPO can arise from the need for aspecialized storage systems. 5 High costs in implementing FEPO can arise from the need for aspecialized storage systems. 6 High costs in implementing FEPO can arise from the need for aspecialized storage systems. 7 Ineffective supplier management results in inaccurate data and reliability. 1 Ineffective supplier management leads to discuption in system maintenance. 9 Ineffective supplier management leads to discuption in system maintenance. 10 The management of suppliers for effective FEPO implementation is increlated with consumer demand fluctuations. 11 The management of suppliers for effective FEPO implementation is increlated with consumer demand fluctuations. 13 Integration of FEPO in existing system maintenance. 14 Inventory rotation relies upon Integration of FEPO in existing systems. 15 Consumer demand fluctuations challenge the integration of FEPO in existing systems. 16 Technological requirements for Inventory and Expiration of FEPO in existing systems. 17 Understanding consumer behaviour is essential for integration of FEPO in existing systems. 18 Management of suppliers of effective FEPO implementati	ID	Description
High costs in implementing FEFO are affected due to the need for a/vanced technology required for inventory and expiration date tracking. High costs in implementing FEFO are amplified by the ongoing expenses of system maintenance. Ineffective supplier management is responsible for incoststent demand from customers. Ineffective supplier management is responsible for incoststent demand from customers. Ineffective supplier management in responsible for incoststent demand from customers. Ineffective supplier management fuels to disruptions in system maintenance. Integrative supplier management fuels to disruptions in system maintenance. Integration of FEFO in existing systems is crucial for system maintenance. Integration of FEFO in existing system is crucial for system maintenance. Integration of FEFO in existing system is crucial for system maintenance. Integration of FEFO in existing systems is crucial for system maintenance. Integration or EEFO in existing systems is crucial for system surving the integration of FEFO in existing systems. Indestanding consume behaviour is essential for integration of FEFO in existing systems. High costs in implementing FEFO indevice fEFO is viral for ensisting systems. Indestanding consume behaviour is essential for integration of FEFO in existing systems. Integration of FEFO in existing system system sort in wattory and Expiration of FEFO in existing systems. Indectore inventory relies on a	1	High costs in implementing FEFO are responsible for inconsistent demand from consumers.
4 High costs in implementing FFD can arise from the need for specialized storage systems. Interfective supplier management results in inaccurate data and reliability. 5 High costs in implementing FFD can arise from the need for specialized storage systems. 6 Inteffective supplier management leads to discuptions in system maintenance. 7 Inteffective supplier management leads to discuptions in system maintenance. 9 Inteffective supplier management leads to discuptions in system maintenance. 9 Interfactive supplier specific type FFD inplementation leads to difficulties in required technological advancements in Inventory and Expiration Date tracting and intrustations challenge the integration of FFD in existing system impedes effective storage management. 11 Integration of FED in existing system since call for system saintenance. 12 Integration of FED in existing systems is crucial for system sintenance. 13 Integration of FED in existing systems is crucial for system sintenance. 14 Inventory rotation relies on for New tory and Expiration of FED in existing systems. 15 Consumer demand fluctuations challenge the integration of FED in existing systems. 16 Technological requirements for Inventory and Expiration of FED in existing systems. 17 Understanding consumer demand fluctuations systemase for inventory and expiration of FED in existin	2	High costs in implementing FEFO affect data accuracy and reliability.
Itp Itp Itp	3	High costs in implementing FEFO are affected due to the need for advanced technology required for inventory and expiration date tracking.
ineffective suppler management results in inaccurate data and reliability. ineffective suppler management is responsible for inconsistent demand from customers. ineffective suppler management affects the storage management for FEPO. The management of suppliers for effective FEPO implementation is interrelated with consumer demand fluctuations. The management of suppliers for effective FEPO implementation is interrelated with consumer demand fluctuations. Integration of FEPO in existing system inpidees effective storage management. Integration of FEPO in existing systems is crucial for system maintenance. Integration of FEPO in existing system is crucial for system maintenance. Integration of FEPO in existing systems. Consumer demand fluctuations challenge the integration of FEPO in existing systems. Management of suppliers for effective FEPO is viral for ensuring systems. Management of suppliers for effective FEPO is viral for ensuring systems. Management of suppliers for effective FEPO in viral on of FEPO in existing systems. Data accuracy and reliability ar ecucial for integration of FEPO in existing systems. Inference in the system sing in integration of FEPO in existing systems. Consumer demand fluctuations impact inventory rotation. System maintenance is essential for inventory rotation. System maintenance is essential for inventory rotation. Syste	4	High costs in implementing FEFO can arise from the need for specialized storage systems.
Ineffective supplier management is responsible for inconsistent demand from customers.Ineffective supplier management leads to disruptions in system maintenance.Ineffective supplier management affects the storage management for FEFO.The management of suppliers for effective FEFO implementation is interrelated with consumer demand fluctuations.Integration of FEFO in existing system impedes effective storage management.Integration of FEFO in existing system sis crucial for system maintenance.Integration of FEFO in existing system sis crucial for system maintenance.Integration of FEFO in existing system sis crucial for system maintenance.Integration of EFO in existing system inpedes effective storage management.Integration of EFO in existing systems is crucial for system maintenance.Integration of EFO in existing systems.Consumer demand fluctuations challenge the integration of FEFO in existing systems.Integration of EFO in existing system.Integration of EFO in existing systems.Integration of EFO in existing systems.Integration of EFO in existing systems.Integration of EFO in existing system.Integration of EFO in existing system.Integration of EFO in existing systems.Integration of EFO in existing system in predeses	5	High costs in implementing FEFO are amplified by the ongoing expenses of system maintenance.
Ineffective supplier management leads in disruptions in system maintenance. Ineffective supplier management affects the storage management for FEPO. Ineffective supplier management affects the storage management is interrelated with consumer demand fluctuations. The management of suppliers for effective FEPO implementation is interrelated with consumer demand fluctuations. Integration of FEPO in existing system impledes effective storage management. Integration of FEPO in existing system impledes effective storage management. Integration of FEPO in existing system is crucial for system maintenance. Integration of FEPO in existing systems is crucial for system maintenance. Consumer demand fluctuations challenge the integration of FEPO in existing systems. Consumer demand fluctuations challenge the integration of FEPO in existing systems. Management of suppliers for effective FEPO is vital for ensuring the integration of FEPO in existing systems. Management of suppliers for effective FEPO is vital for ensuring the integration of FEPO in existing systems. Management of suppliers on advanced tracking systems for inventory and expirit and expiry dates. Data accuracy and reliability are existential for inventory rotation. Management of suppliers for effective FEPO implementation influences inventory rotation. System maintenance is essential for inventory rotation. Management of suppliers for effective FEPO implementation influences inventory rotation. Consumer demand fluctuations impact inventory rotation. Management of suppli	5	Ineffective supplier management results in inaccurate data and reliability.
Ineffective supplier management affects the storage management for FEFO. Integration of FEFO in existing system impedes effective FEFO implementation leads to difficulties in required technological advancements in Inventory and Expiration Date tra Integration of FEFO in existing system is crucial for system maintenance. Integration of FEFO in existing system is crucial for system maintenance. Integration of FEFO in existing system is crucial for system maintenance. Integration of FEFO in existing systems is crucial for system maintenance. Integration of FEFO in existing systems. Consumer demand fluctuations challenge the integration of FEFO in existing systems. Understanding consumer behaviour is essential for integration of FEFO in existing systems. Integration of FEFO in existing system in order the integration of FEFO in existing systems. Data accuracy and reliability are crucial for integration of FEFO in existing systems. Effective inventory roles on advanced tracking systems for inventory and expiry dates. Data accuracy and reliability are essential for inventory rotation. System maintenance is essential for inventory rotation. System maintenance is essential for inventory rotation. Inventory rotation elevates the costs in implementation fFEFO. Consumer demand fluctuations lead to challenges in effective storage management. Inventory rotation enhances operational efficiency.	7	Ineffective supplier management is responsible for inconsistent demand from customers.
10 The management of suppliers for effective FEFO implementation is interrelated with consumer demand fluctuations. 11 The management of suppliers for effective FEFO implementation leads to difficulties in required technological advancements in Inventory and Expiration Date trad 13 Integration of FEFO in existing systems is crucial for system maintenance. 14 Inventory rotation relies upon Integration of FEFO in existing systems. 15 Consumer demand fluctuations challenge the integration of FEFO in existing systems. 16 Technological requirements for Inventory and Expiration of FEFO in existing systems. 16 Management of suppliers for effective FEFO is vital for ensuring the integration of FEFO in existing systems. 17 Understanding consumer behaviour is essential for integration of FEFO in existing systems. 18 Management of suppliers for effective FEFO is vital for ensuring the integration of FEFO in existing systems. 18 Effective inventory rotation of FEFO in existing systems. 19 Eat accuracy and reliability are crucial for integration of FEFO in existing systems. 10 High costs in implementing FEFO hinder the integration of FEFO in existing systems. 10 Intertory orbiton benefits effective storage management. 11 Inventory rotation benefits effective storage management. 12 Management of suppliers f	3	Ineffective supplier management leads to disruptions in system maintenance.
11 The management of suppliers for effective FEFO implementation leads to difficulties in required technological advancements in Inventory and Expiration Date trading system is grucial for system maintenance. 12 Integration of FEFO in existing systems is crucial for system maintenance. 14 Inventory rotation relies upon Integration of FEFO in existing systems. 15 Consumer demand fluctuations challenge the integration of FEFO in existing systems. 16 Technological requirements for Inventory and Expiration Date tracking are essential for enabling integration of FEFO in existing systems. 17 Understanding consumer behaviour is essential for integration of FEFO in existing systems. 18 Management of suppliers for effective FEFO is virtal for ensuring the integration of FEFO in existing systems. 19 Data accuracy and reliability are crucial for integration of FEFO in existing systems. 210 High costs in implementing FEFO hinder the integration of FEFO. 221 Data accuracy and reliability are essential for inventory rotation. 222 Data accuracy and reliability are essential for inventory rotation. 223 System maintenance is essential for inventory rotation. 224 Consumer demand fluctuations impact inventory rotation. 225 System maintenance is essential for inventory rotations. 236 Consumer behavioural patterns	9	Ineffective supplier management affects the storage management for FEFO.
12Integration of FEPO in existing system impedes effective storage management.13Integration of FEPO in existing systems is crucial for system maintenance.14Inventory rotation relies upon Integration of FEPO in existing systems.15Consumer demand fluctuations challenge the integration of FEPO in existing systems.16Technological requirements for Inventory and Expiration Date tracking are essential for enabling integration of FEPO in existing systems.17Understanding consumer behaviour is essential for integration of FEPO in existing systems.18Management of suppliers for effective FEPO is vital for ensuring the integration of FEPO in existing systems.20High costs in implementing FEPO hinder the integration of FEPO in existing systems.21Effective inventory relies on advanced tracking systems for inventory and expiry dates.22Data accuracy and reliability are costent for FEPO.23Inventory rotation benefits effective storage management for FEPO.24Consumer demand fluctuations impact inventory rotation.25System maintenance is essential for inventory rotation.26Inventory rotation enlarces operational efficiency.27Management of suppliers for effective FEPO implementation influences inventory rotation.28Inventory rotation enlarces operational efficiency.29Consumer demand fluctuations in implementing FEPO.20Challenges in effective storage management for storage management.30Challenges in effective storage management infective storage management.31Consumer demand fluctuations callenges in effective s	10	The management of suppliers for effective FEFO implementation is interrelated with consumer demand fluctuations.
13Integration of FEFO in existing systems is crucial for system maintenance.14Inventory rotation relies upon Integration of FEFO in existing systems.15Consumer demand fluctuations challenge the integration of FEFO in existing systems.16Technological requirements for Inventory and Expiration Date tracking are essential for inegration of FEFO in existing systems.16Management of suppliers for effective FEFO is vital for integration of FEFO in existing systems.17Understanding consumer behaviour is essential for integration of FEFO in existing systems.18Management of suppliers for effective FEFO is vital for neutring systems.19Data accuracy and reliability are crucial for integration of FEFO in existing systems.21Effective inventory reles on advanced tracking systems for inventory and expiry dates.22Data accuracy and reliability are essential for inventory rotation.23Inventory rotation benefits effective storage management for FEFO.24Consumer demand fluctuations impact inventory rotation.25System maintenance is essential for inventory rotation.26Inventory rotation elevates the costs in implementing FEFO.27Management of suppliers for effective FEFO linghementation influences inventory rotation.28Consumer demand fluctuations lead to challenges in effective storage management.29Consumer demand fluctuations lead to challenges in effective storage management.30Challenges in effective storage management for EFO leavier with regulations.31Unreliable and inaccurate data leads to challenges in effective storage management.<	11	The management of suppliers for effective FEFO implementation leads to difficulties in required technological advancements in Inventory and Expiration Date tracking
14Inventory rotation relies upon Integration of FEFO in existing systems.15Consumer demand fluctuations challenge the integration of FEFO in existing systems.16Technological requirements for Inventory and Expiration Date tracking are essential for enabling integration of FEFO in existing systems.17Understanding consumer behaviour is essential for integration of FEFO in existing systems.18Management of suppliers for effective FEFO is vital for ensuring the integration of FEFO in existing systems.18High costs in implementing FEFO hinder the integration of FEFO in existing systems.20High costs in implementing FEFO hinder the integration of FEFO in existing systems.21Effective inventory rolation advanced tracking systems for inventory and expiry dates.22Data accuracy and reliability are essential for inventory rotation.23Inventory rotation benefits effective storage management for FEFO.24Consumer demand fluctuations implementing FEFO.25System maintenance is essential for inventory rotation.26Inventory rotation enhances operational efficiency.27Consumer behavioural patterns help optimize inventory rotations.28Consumer behavioural patterns help optimize infective storage management.29Consumer demand fluctuations lengagement lead to high costs in implementing FEFO.31Challenges in effective storage management.32Unreliable and inaccurate data leads to challenges in effective storage management.33Seasonal availability and supply variations complicate effective storage management.34Seasonal ava	12	Integration of FEFO in existing system impedes effective storage management.
15Consumer demand fluctuations challenge the integration of FEFO in existing systems.16Technological requirements for Inventory and Expiration Date tracking are essential for enabling integration of FEFO in existing systems.18Management of suppliers for effective FEFO is vital for ensuring of FEFO in existing systems.19Data accuracy and reliability are crucial for integration of FEFO in existing systems.10Effective inventory relies on advanced tracking system for inventory and expiry dates.21Data accuracy and reliability are sesential for inventory rolation.23Inventory rolation benefits effective storage management for FEFO.24Consumer demand fluctuations impact inventory rotation.25System maintenance is essential for inventory rotation.26Inventory rotation elevates the costs in implementing FEFO.27Management of suppliers for effective FEFO in pulsemention influences inventory rotation.28System maintenance is essential for inventory rotation.29Consumer behavioural patterns help optimize inventory rotations.20Consumer behavioural patterns help optimize inventory rotations.21Consumer demand fluctuations lead to challenges in effective storage management.23Unreliable and inaccurate data leads to challenges in effective storage management.23Consumer demand fluctuations lead to challenges in effective storage management.24Consumer demand fluctuations complicate effective storage management.25Seasonal availability and supply variations complicate effective storage management.26Consumer deha	13	Integration of FEFO in existing systems is crucial for system maintenance.
16Technological requirements for Inventory and Expiration Date tracking are essential for enabling integration of FEFO in existing systems.17Understanding consumer behaviour is essential for integration of FEFO in existing systems.18Management of suppliers for effective FEFO is vital for ensuring the integration of FEFO in existing systems.19Data accuracy and reliability are crucial for integration of FEFO in existing systems.10Effective inventory relies on advanced tracking systems for inventory rotation.12Effective inventory relies on advanced tracking systems for inventory and expiry dates.13Inventory rotation benefits effective storage management for FEFO.14Consumer demand fluctuations impact inventory rotation.15System maintenance is essential for inventory rotation.16Inventory rotation enhances operational efficiency.17Management of suppliers for effective FEFO implementation influences inventory rotation.18Consumer behavioural patterns help optimize inventory rotations.19Consumer behavioural patterns help optimize inventory rotations.10Consumer behavioural patterns help optimize inventory rotations.11Consumer behavioural patterns help optimize inventory rotations.13Challenges in effective storage management.14Challenges in effective storage management affect operational efficiency.15Seasonal availability and supply variations complicate effective storage management.16Consumer behaviour and purchasing patterns affect overstorage management.17Challenges in effective storage mana	14	Inventory rotation relies upon Integration of FEFO in existing systems.
 Understanding consumer behaviour is essential for integration of FEPO in existing systems. Management of suppliers for effective FEPO is vital for ensuring the integration of FEPO in existing systems. Data accuracy and reliability are crucial for integration of FEPO in existing systems. High costs in implementing FEPO hinder the integration of FEPO in existing systems. Effective inventory relies on advanced tracking systems for inventory and expiry dates. Data accuracy and reliability are essential for inventory rotation. Inventory rotation benefits effective storage management for FEPO. Consumer demand fluctuations impact inventory rotation. Inventory rotation elevates the costs in implementing FEPO. Management of suppliers for effective FEPO implementation influences inventory rotation. Inventory rotation enhances operational efficiency. Consumer demand fluctuations elay to thigh costs in implementing FEPO. Consumer behavioural patterns help optimize inventory rotations. Challenges in effective storage management lead to high costs in implementing FEPO. Challenges in effective storage management affect operational efficiency. Challenges in effective storage management affect operational efficiency. Challenges in effective storage management affect operational efficiency. Seasonal availability and supply variations complicate effective storage management. Effective storage management effective storage management. Effective storage management affect operational efficiency. Seasonal availability and supply variations complicate effective storage management. Effective storage management elevates the high costs in implementing FEPO. Seasonal availability and supply variation lead to noncompliance with regulations. Effective storage management elevates the high costs in implementing FEPO. Seaso	15	Consumer demand fluctuations challenge the integration of FEFO in existing systems.
18Management of suppliers for effective FEFO is vital for ensuring the integration of FEFO in existing systems.19Data accuracy and reliability are crucial for integration of FEFO in existing systems.11Effective inventory relies on advanced tracking systems for inventory and expiry dates.21Effective inventory relies on advanced tracking systems for inventory and expiry dates.21Data accuracy and reliability are essential for inventory rotation.21Inventory rotation benefits effective storage management for FEFO.24Consumer demand fluctuations impact inventory rotation.25System maintenance is essential for inventory rotation.26Inventory rotation elevates the costs in implementing FEFO.27Management of suppliers for effective FEFO implementation influences inventory rotation.28Inventory rotation enhances operational efficiency.29Consumer behavioural patterns help optimize inventory rotations.30Challenges in effective storage management for SEFO leads to management.31Unreliable and inaccurate data leads to challenges in effective storage management.33Challenges in effective storage management affect operational efficiency.34Consumer behavioural patterns affect operational efficiency.35Seasonal availability and supply variations complicate effective storage management.36Challenges in effective storage management for FEFO lead to noncompliance with regulations.36Challenges in effective storage management for FEFO lead to noncompliance with regulations.37Consumer behaviour and purchasing patter	16	Technological requirements for Inventory and Expiration Date tracking are essential for enabling integration of FEFO in existing systems.
19Data accuracy and reliability are crucial for integration of FEFO in existing systems.20High costs in implementing FEFO hinder the integration of FEFO in existing systems.21Effective inventory relies on advanced tracking systems for inventory and expiry dates.22Data accuracy and reliability are essential for inventory rotation.23Inventory rotation benefits effective storage management for FEFO.24Consumer demand fluctuations impact inventory rotation.25System maintenance is essential for inventory rotation.26Inventory rotation elevates the costs in implementing FEFO.27Management of suppliers for effective FEFO implementation influences inventory rotation.28Inventory rotation enhances operational efficiency.29Consumer behavioural patterns help optimize inventory rotations.30Challenges in effective storage management lead to high costs in implementing FEFO.31Curreliable and inaccurate data leads to challenges in effective storage management.32Unreliables an infective storage management for FEFO lead to noncompliance with regulations.33Challenges in effective storage management.34Effective storage management levates the high costs in implementing FEFO.35Seasonal availability and supply variations complicate effective storage management.36Effective storage management for FEFO lead to noncompliance with regulations.36Seasonal availability and supply variations affect storage management.37Consumer behaviour and purchasing patterns affect storage management.38Imp	17	Understanding consumer behaviour is essential for integration of FEFO in existing systems.
 High costs in implementing FEFO hinder the integration of FEFO in existing systems. Effective inventory relies on advanced tracking systems for inventory and expiry dates. Data accuracy and reliability are essential for inventory rotation. Inventory rotation benefits effective storage management for FEFO. Consumer demand fluctuations impact inventory rotation. System maintenance is essential for inventory rotation. System maintenance is essential for inventory rotation. Management of suppliers for effective FEFO implementation influences inventory rotation. Inventory rotation elevates the costs in implementing FEFO. Consumer demand fluctuations lead to challenges in effective storage management. Consumer demand fluctuations lead to challenges in effective storage management. Challenges in effective storage management affect operational efficiency. Seasonal availability and supply variations complicate effective storage management. Challenges in effective storage management affect operational efficiency. Seasonal availability and supply variations complicate effective storage management. Effective storage management affect operational efficiency. Seasonal availability and supply variation complicate effective storage management. Improper system maintenance hinders the ability to have effective storage management. Seasonal availability and supply variation cate costs an implementing FEFO. Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation cate costs in implementing FEFO. Advanced technology for inventory and expiration date tracking is essential for management. Precise and trustworthy data is essential for predictive storage management. Consumer behaviour and purchasing patterns affect storage management. Precise and trus	18	Management of suppliers for effective FEFO is vital for ensuring the integration of FEFO in existing systems.
21Effective inventory relies on advanced tracking systems for inventory and expiry dates.22Data accuracy and reliability are essential for inventory rotation.23Inventory rotation benefits effective storage management for FEFO.24Consumer demand fluctuations inpact inventory rotation.25System maintenance is essential for inventory rotation.26Inventory rotation elevates the costs in implementing FEFO.27Management of suppliers for effective FEFO implementation influences inventory rotation.28Inventory rotation enhances operational efficiency.29Consumer behavioural patterns help optimize inventory rotations.31Challenges in effective storage management lead to high costs in implementing FEFO.32Unreliable and inaccurate data leads to challenges in effective storage management.32Unreliable and infective storage management for FEFO lead to noncompliance with regulations.34Challenges in effective storage management affect operational efficiency.35Seasonal availability and supply variations complicate effective storage management.36Effective storage management affect operational efficiency.35Seasonal availability and supply variations complicate effective storage management.36Improper system maintenance hinders the ability to have effective storage management.37Consumer behaviour and purchasing patterns affect torage management.38Improper system maintenance hinders the ability to have effective storage management.39Seasonal availability and supply variation lead to increased expenses in impleme	19	Data accuracy and reliability are crucial for integration of FEFO in existing systems.
 Data accuracy and reliability are essential for inventory rotation. Inventory rotation benefits effective storage management for FEFO. Consumer demand fluctuations impact inventory rotation. System maintenance is essential for inventory rotation. Inventory rotation elevates the costs in implementing FEFO. Management of suppliers for effective FEFO implementation influences inventory rotation. Inventory rotation enhances operational efficiency. Consumer behavioural patterns help optimize inventory rotations. Challenges in effective storage management lead to high costs in implementing FEFO. Consumer demand fluctuations lead to challenges in effective storage management. Unreliable and inaccurate data leads to challenges in effective storage management. Challenges in effective storage management for FEFO lead to noncompliance with regulations. Challenges in effective storage management affect operational efficiency. Seasonal availability and supply variations complicate effective storage management. Effective storage management effect storage management. Effective storage management effect storage management. Effective storage management effect storage management. Seasonal availability and supply variations complicate effective storage management. Seasonal availability and supply variation complicate effective storage management. Seasonal availability and supply variation in the average dependent. Seasonal availability and supply variation date tracking is essential for managing seasonal availability and supply variation. Accurate tracking technology is are crucial for adapting to changing demand patterns. Precise and trustworthy data is essential for redicting demand patterns. Unpredictable demand can lead to verstocking or stockouts. Consumer demand fluctuations are interrelated with consum	20	High costs in implementing FEFO hinder the integration of FEFO in existing systems.
 Inventory rotation benefits effective storage management for FEFO. Consumer demand fluctuations impact inventory rotation. System maintenance is essential for inventory rotation. Inventory rotation elevates the costs in implementing FEFO. Inventory rotation enhances operational efficiency. Consumer behavioural patterns help optimize inventory rotations. Consumer beravioural patterns help optimize inventory rotations. Consumer behavioural patterns help optimize inventory rotations. Consumer behavioural patterns help optimize inventory rotations. Consumer behavioural patterns help optimize inventory rotations. Consumer demand fluctuations lead to challenges in effective storage management. Unreliable and inaccurate data leads to challenges in effective storage management. Challenges in effective storage management affect operational efficiency. Seasonal availability and supply variations complicate effective storage management. Effective storage management elevates the high costs in implementing FEFO. Consumer behaviour and purchasing patterns affect storage management. Effective storage management elevates the high costs in implementing FEFO. Consumer behaviour and purchasing patterns affect storage management. Effective storage management elevates the high costs in implementing FEFO. Consumer behaviour and purchasing patterns affect storage management. Seasonal availability and supply variation lead to increased expenses in implementing FEFO. Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation. Accurate tracking technologies are crucial for adapting to changing demand patterns. Precise and trustworthy data is essential for predicting demand patterns. Unpredictable demand can lead to overstocking or stockouts.<td></td><td>Effective inventory relies on advanced tracking systems for inventory and expiry dates.</td>		Effective inventory relies on advanced tracking systems for inventory and expiry dates.
 Consumer demand fluctuations impact inventory rotation. System maintenance is essential for inventory rotation. Inventory rotation elevates the costs in implementing FEFO. Management of suppliers for effective FEFO implementation influences inventory rotation. Inventory rotation enhances operational efficiency. Consumer behavioural patterns help optimize inventory rotations. Challenges in effective storage management lead to high costs in implementing FEFO. Consumer demand fluctuations lead to challenges in effective storage management. Unreliable and inaccurate data leads to challenges in effective storage management. Challenges in effective storage management for FEFO lead to noncompliance with regulations. Challenges in effective storage management for FEFO lead to noncompliance with regulations. Challenges in effective storage management for version al efficiency. Seasonal availability and supply variations complicate effective storage management. Effective storage management elevates the high costs in implementing FEFO. Consumer behaviour and purchasing patterns affect storage management. Effective storage management elevates the ability to have effective storage management. Seasonal availability and supply variation lead to increased expenses in implementing FEFO. Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation. Accurate tracking technologies are crucial for adapting to changing demand patterns. Precise and trustworthy data is essential for predicting or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 	22	Data accuracy and reliability are essential for inventory rotation.
25System maintenance is essential for inventory rotation.26Inventory rotation elevates the costs in implementing FEFO.27Management of suppliers for effective FEFO implementation influences inventory rotation.28Inventory rotation enhances operational efficiency.29Consumer behavioural patterns help optimize inventory rotations.30Challenges in effective storage management lead to high costs in implementing FEFO.31Consumer demand fluctuations lead to challenges in effective storage management.32Unreliable and inaccurate data leads to challenges in effective storage management.33Challenges in effective storage management of FEFO lead to noncompliance with regulations.34Challenges in effective storage management affect operational efficiency.35Seasonal availability and supply variations complicate effective storage management.36Effective storage management effect storage management.36Effective storage management affect operational efficiency.37Seasonal availability and supply variations complicate effective storage management.38Improper system maintenance hinders the ability to have effective storage management.39Seasonal availability and supply variation lead to increased expenses in implementing FEFO.30Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation.31Accurate tracking technologies are crucial for adapting to changing demand patterns.32Precise and trustworthy data is essential for predicting demand patterns.33 <td>23</td> <td>Inventory rotation benefits effective storage management for FEFO.</td>	23	Inventory rotation benefits effective storage management for FEFO.
 Inventory rotation elevates the costs in implementing FEFO. Management of suppliers for effective FEFO implementation influences inventory rotation. Inventory rotation enhances operational efficiency. Consumer behavioural patterns help optimize inventory rotations. Challenges in effective storage management lead to high costs in implementing FEFO. Consumer demand fluctuations lead to challenges in effective storage management. Unreliable and inaccurate data leads to challenges in effective storage management. Challenges in effective storage management for FEFO lead to noncompliance with regulations. Challenges in effective storage management affect operational efficiency. Seasonal availability and supply variations complicate effective storage management. Consumer behaviour and purchasing patterns affect storage management. Consumer behaviour and purchasing patterns affect storage management. Improper system maintenance hinders the ability to have effective storage management. Seasonal availability and supply variation lead to increased expenses in implementing FEFO. Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation for adapting to changing demand patterns. Precise and trustworthy data is essential for predicting demand patterns. Unpredictable demand can lead to overstockung or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 	24	Consumer demand fluctuations impact inventory rotation.
 Management of suppliers for effective FEFO implementation influences inventory rotation. Inventory rotation enhances operational efficiency. Consumer behavioural patterns help optimize inventory rotations. Challenges in effective storage management lead to high costs in implementing FEFO. Consumer demand fluctuations lead to challenges in effective storage management. Unreliable and inaccurate data leads to challenges in effective storage management. Challenges in effective storage management for FEFO lead to noncompliance with regulations. Challenges in effective storage management affect operational efficiency. Seasonal availability and supply variations complicate effective storage management. Effective storage management elevates the high costs in implementing FEFO. Consumer behaviour and purchasing patterns affect storage management. Improper system maintenance hinders the ability to have effective storage management. Seasonal availability and supply variation lead to increased expenses in implementing FEFO. Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation. Accurate tracking technologies are crucial for adapting to changing demand patterns. Precise and trustworthy data is essential for predicting demand patterns. Unpredictable demand can lead to overstocking or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 		System maintenance is essential for inventory rotation.
28Invertory rotation enhances operational efficiency.29Consumer behavioural patterns help optimize inventory rotations.30Challenges in effective storage management lead to high costs in implementing FEFO.31Consumer demand fluctuations lead to challenges in effective storage management.32Unreliable and inaccurate data leads to challenges in effective storage management.33Challenges in effective storage management affect operational efficiency.34Challenges in effective storage management affect operational efficiency.35Seasonal availability and supply variations complicate effective storage management.36Effective storage management elevates the high costs in implementing FEFO.37Consumer behaviour and purchasing patterns affect storage management.38Improper system maintenance hinders the ability to have effective storage management.39Seasonal availability and supply variation lead to increased expenses in implementing FEFO.40Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation lead to changing demand patterns.41Accurate tracking technologies are crucial for adapting to changing demand patterns.42Precise and trustworthy data is essential for predicting demand patterns.43Unpredictable demand can lead to overstocking or stockouts.44Consumer demand fluctuations are interrelated with consumer behaviour.	26	Inventory rotation elevates the costs in implementing FEFO.
 Consumer behavioural patterns help optimize inventory rotations. Challenges in effective storage management lead to high costs in implementing FEFO. Consumer demand fluctuations lead to challenges in effective storage management. Unreliable and inaccurate data leads to challenges in effective storage management. Challenges in effective storage management for FEFO lead to noncompliance with regulations. Challenges in effective storage management affect operational efficiency. Seasonal availability and supply variations complicate effective storage management. Effective storage management elevates the high costs in implementing FEFO. Consumer behaviour and purchasing patterns affect storage management. Improper system maintenance hinders the ability to have effective storage management. Seasonal availability and supply variation lead to increased expenses in implementing FEFO. Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation for changing demand patterns. Precise and trustworthy data is essential for predicting demand patterns. Unpredictable demand can lead to overstocking or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 	27	Management of suppliers for effective FEFO implementation influences inventory rotation.
 Challenges in effective storage management lead to high costs in implementing FEFO. Consumer demand fluctuations lead to challenges in effective storage management. Unreliable and inaccurate data leads to challenges in effective storage management. Challenges in effective storage management for FEFO lead to noncompliance with regulations. Challenges in effective storage management for FEFO lead to noncompliance with regulations. Challenges in effective storage management affect operational efficiency. Seasonal availability and supply variations complicate effective storage management. Effective storage management elevates the high costs in implementing FEFO. Consumer behaviour and purchasing patterns affect storage management. Improper system maintenance hinders the ability to have effective storage management. Seasonal availability and supply variation lead to increased expenses in implementing FEFO. Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation. Accurate tracking technologies are crucial for adapting to changing demand patterns. Precise and trustworthy data is essential for predicting demand patterns. Unpredictable demand can lead to overstocking or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 	28	Inventory rotation enhances operational efficiency.
 Consumer demand fluctuations lead to challenges in effective storage management. Unreliable and inaccurate data leads to challenges in effective storage management. Challenges in effective storage management for FEFO lead to noncompliance with regulations. Challenges in effective storage management affect operational efficiency. Seasonal availability and supply variations complicate effective storage management. Effective storage management elevates the high costs in implementing FEFO. Consumer behaviour and purchasing patterns affect storage management. Improper system maintenance hinders the ability to have effective storage management. Seasonal availability and supply variation lead to increased expenses in implementing FEFO. Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation. Accurate tracking technologies are crucial for adapting to changing demand patterns. Precise and trustworthy data is essential for predicting demand patterns. Unpredictable demand can lead to overstocking or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 	29	Consumer behavioural patterns help optimize inventory rotations.
 Unreliable and inaccurate data leads to challenges in effective storage management. Challenges in effective storage management for FEFO lead to noncompliance with regulations. Challenges in effective storage management affect operational efficiency. Seasonal availability and supply variations complicate effective storage management. Effective storage management elevates the high costs in implementing FEFO. Consumer behaviour and purchasing patterns affect storage management. Seasonal availability and supply variation lead to increased expenses in implementing FEFO. Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation. Accurate tracking technologies are crucial for adapting to changing demand patterns. Precise and trustworthy data is essential for predicting demand patterns. Unpredictable demand can lead to overstocking or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 		
 Challenges in effective storage management for FEFO lead to noncompliance with regulations. Challenges in effective storage management affect operational efficiency. Seasonal availability and supply variations complicate effective storage management. Effective storage management elevates the high costs in implementing FEFO. Consumer behaviour and purchasing patterns affect storage management. Improper system maintenance hinders the ability to have effective storage management. Seasonal availability and supply variation lead to increased expenses in implementing FEFO. Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation. Accurate tracking technologies are crucial for adapting to changing demand patterns. Precise and trustworthy data is essential for predicting demand patterns. Unpredictable demand can lead to overstocking or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 		
 Challenges in effective storage management affect operational efficiency. Seasonal availability and supply variations complicate effective storage management. Effective storage management elevates the high costs in implementing FEFO. Consumer behaviour and purchasing patterns affect storage management. Improper system maintenance hinders the ability to have effective storage management. Seasonal availability and supply variation lead to increased expenses in implementing FEFO. Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation. Accurate tracking technologies are crucial for adapting to changing demand patterns. Precise and trustworthy data is essential for predicting demand patterns. Unpredictable demand can lead to overstocking or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 		
 Seasonal availability and supply variations complicate effective storage management. Effective storage management elevates the high costs in implementing FEFO. Consumer behaviour and purchasing patterns affect storage management. Improper system maintenance hinders the ability to have effective storage management. Seasonal availability and supply variation lead to increased expenses in implementing FEFO. Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation. Accurate tracking technologies are crucial for adapting to changing demand patterns. Precise and trustworthy data is essential for predicting demand patterns. Unpredictable demand can lead to overstocking or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 		
 Effective storage management elevates the high costs in implementing FEFO. Consumer behaviour and purchasing patterns affect storage management. Improper system maintenance hinders the ability to have effective storage management. Seasonal availability and supply variation lead to increased expenses in implementing FEFO. Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation. Accurate tracking technologies are crucial for adapting to changing demand patterns. Precise and trustworthy data is essential for predicting demand patterns. Unpredictable demand can lead to overstocking or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 		
 Consumer behaviour and purchasing patterns affect storage management. Improper system maintenance hinders the ability to have effective storage management. Seasonal availability and supply variation lead to increased expenses in implementing FEFO. Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation. Accurate tracking technologies are crucial for adapting to changing demand patterns. Precise and trustworthy data is essential for predicting demand patterns. Unpredictable demand can lead to overstocking or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 		
 Improper system maintenance hinders the ability to have effective storage management. Seasonal availability and supply variation lead to increased expenses in implementing FEFO. Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation. Accurate tracking technologies are crucial for adapting to changing demand patterns. Precise and trustworthy data is essential for predicting demand patterns. Unpredictable demand can lead to overstocking or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 		
 Seasonal availability and supply variation lead to increased expenses in implementing FEFO. Advanced technology for inventory and expiration date tracking is essential for managing seasonal availability and supply variation. Accurate tracking technologies are crucial for adapting to changing demand patterns. Precise and trustworthy data is essential for predicting demand patterns. Unpredictable demand can lead to overstocking or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 		
 Advanced technology for invertory and expiration date tracking is essential for managing seasonal availability and supply variation. Accurate tracking technologies are crucial for adapting to changing demand patterns. Precise and trustworthy data is essential for predicting demand patterns. Unpredictable demand can lead to overstocking or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 		
 Accurate tracking technologies are crucial for adapting to changing demand patterns. Precise and trustworthy data is essential for predicting demand patterns. Unpredictable demand can lead to overstocking or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 		
 Precise and trustworthy data is essential for predicting demand patterns. Unpredictable demand can lead to overstocking or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 		
 Unpredictable demand can lead to overstocking or stockouts. Consumer demand fluctuations are interrelated with consumer behaviour. 		
44 Consumer demand fluctuations are interrelated with consumer behaviour.		
45 Understanding consumer preferences and purchasing patterns can drive the need for advanced tracking systems.	44	
5 I IIII 01 III 01 III 01 III 01 III 0 IIII 0 III 0 IIII 0 III 0 IIII 0 IIII 0 IIII 0 IIII 0 IIII 0 IIII 0 III 0 III 0 I	45	Understanding consumer preferences and purchasing patterns can drive the need for advanced tracking systems.

Table A1 (continued)

ID	Description
I46	Understanding consumer preferences and expectations can inform compliance strategies.
I47	Adhering to regulations often requires significant investments in staff training and technology.
I48	Changes in consumer preferences may necessitate updates to compliance standards.
I49	Compliance standards often dictate the need for specific tracking technologies.
150	Regular upkeep of inventory management systems is essential to ensure they function correctly and meet compliance standards.
I51	Data accuracy and reliability are important for regular compliance and training.
152	Advanced tracking technologies streamline inventory management processes while reducing manual errors.
153	Accurate and reliable data is essential for operational efficiency.
I54	Regular maintenance of systems ensures they function optimally and reduces downtime.
155	Effectively incorporating FEFO practices into current workflows enhances inventory management processes.
156	Effectively responding to changes in demand requires streamlined operations that can adapt quickly, ensuring that inventory levels align with consumer needs.
157	Advanced tracking technologies are essential for capturing precise and accurate data.
158	Regular maintenance of data management systems is essential to ensure that data remains accurate and reliable.
159	Maintaining inventory management systems is crucial for adapting to changing demand patterns.
160	Ongoing maintenance of tracking systems is essential to ensure they operate effectively.
I61	Regular upkeep of systems is vital for preventing errors, ensuring that data is correctly processed and stored.

Table A2

Causal matrix

Notation	Challenges	Sr	Sc	Sr-Sc	$\mathbf{Sr} + \mathbf{Sc}$	Cause/Effect	Ranking
P1	P1	4.7022	5.6315	-0.9293	10.3337	Effect	4
P2	P2	4.6298	4.6897	-0.0599	9.3195	Effect	9
P3	P3	5.3588	4.593	0.7658	9.9518	Cause	5
P4	P4	5.5552	4.2265	1.3287	9.7817	Cause	7
P5	P5	5.7667	5.1642	0.6025	10.9309	Cause	1
P6	P6	3.7838	3.4361	0.3477	7.2199	Cause	13
P7	P7	4.8188	4.8915	-0.0727	9.7103	Effect	8
P8	P8	3.3658	5.1387	-1.7729	8.5045	Effect	12
P9	Р9	4.9496	3.8534	1.0962	8.803	Cause	10
P10	P10	4.4732	4.1406	0.3326	8.6138	Cause	11
P11	P11	4.6312	5.2833	-0.6521	9.9145	Effect	6
P12	P12	5.0962	5.5365	-0.4403	10.6327	Effect	3
P13	P13	5.1387	5.685	-0.5463	10.8237	Effect	2

Table A3 Expe

Expert 1	Matrix										
	C01	C02	C03	C04	C05	C06	C07	C08	C09	C10	C11
C01	1	1	0	0	3	2	3	2	3	1	3
C02	1	1	3	1	4	4	1	4	2	1	1
C03	3	2	1	1	3	3	2	4	0	3	2
C04	3	2	0	1	2	4	4	4	3	1	2
C05	1	1	3	3	1	3	3	0	4	1	0
C06	3	3	1	4	4	1	2	3	4	0	3
C07	1	2	1	4	0	0	1	3	3	1	1
C08	3	3	3	1	0	0	2	1	3	2	4
C09	3	1	2	3	3	2	2	3	1	2	3
C10	0	1	3	2	1	4	3	3	4	1	0
C11	1	3	4	4	4	2	2	0	2	1	1
C12	0	3	1	2	1	2	4	1	4	3	1
C13	3	1	1	1	1	2	1	3	1	1	4

Table A4 Expert 2 Matrix

	C01	C02	C03	C04	C05	C06	C07	C08	C09	C10	C11	C12	C13
C01	1	1	3	4	3	3	0	2	1	2	4	3	0
C02	1	1	4	2	4	2	4	1	3	1	2	1	0
C03	1	0	1	3	4	4	3	0	4	2	0	0	2
C04	3	0	3	1	0	1	0	3	0	0	3	0	2
C05	2	4	0	3	1	2	2	2	2	0	1	4	2
C06	4	0	1	2	0	1	1	1	0	1	0	3	1
C07	2	2	4	4	3	2	1	2	0	2	4	2	2
C08	1	4	0	1	4	2	4	1	3	1	0	1	3
C09	1	1	4	2	4	4	0	2	1	1	0	1	1

(continued on next page)

C12

 C13

2 1

2

Table A4 (continued)

	C01	C02	C03	C04	C05	C06	C07	C08	C09	C10	C11	C12	C13
C10	2	3	2	3	1	1	0	4	1	1	1	4	2
C11	2	4	1	4	1	4	4	4	4	1	1	3	1
C12	3	3	2	4	1	2	3	3	1	4	0	1	2
C13	3	4	2	3	4	0	0	3	3	0	2	1	1

Table A5

Expert 3 Matrix

	C01	C02	C03	C04	C05	C06	C07	C08	C09	C10	C11	C12	C13
C01	1	3	1	4	3	4	0	2	3	2	2	3	2
C02	3	1	2	4	0	1	4	0	2	0	4	1	0
C03	3	0	1	0	0	1	1	4	4	1	0	3	1
C04	0	3	4	1	3	1	1	4	3	1	3	2	1
C05	4	4	4	4	1	4	4	1	1	2	2	4	4
C06	2	1	4	1	2	1	3	4	2	4	0	0	4
C07	0	0	0	2	3	0	1	1	2	1	2	4	2
C08	2	3	1	4	4	4	4	1	2	0	4	1	4
C09	3	4	4	0	3	0	4	1	1	1	0	4	1
C10	0	1	4	1	2	0	4	0	2	1	2	3	1
C11	1	2	0	0	0	4	3	3	3	1	1	2	3
C12	2	4	3	4	2	2	0	3	0	4	4	1	2
C13	3	2	3	2	1	3	2	2	1	3	1	3	1

Table A6

Expert 4 Matrix

	C01	C02	C03	C04	C05	C06	C07	C08	C09	C10	C11	C12	C13
C01	1	3	2	2	4	2	3	4	1	3	0	3	4
C02	2	1	0	2	2	0	4	2	2	1	4	0	1
C03	0	2	1	1	4	4	2	1	3	1	3	3	1
C04	1	4	3	1	2	1	3	1	0	4	0	4	0
C05	2	1	0	1	1	4	4	3	2	0	4	4	1
C06	4	3	1	0	2	1	4	4	4	4	0	0	2
C07	3	2	0	0	4	4	1	3	4	0	2	1	3
C08	2	3	3	1	3	1	4	1	3	4	4	2	1
C09	0	2	4	4	3	3	2	3	1	1	1	2	0
C10	4	2	1	2	0	1	0	2	0	1	4	2	0
C11	4	2	0	0	3	3	4	3	2	0	1	1	0
C12	0	0	4	1	3	3	0	0	1	3	3	1	0
C13	3	1	3	2	0	4	1	4	1	2	1	2	1

Table A7 Expert 5 Matrix

	C01	C02	C03	C04	C05	C06	C07	C08	C09	C10	C11	C12	C13
C01	1	2	0	4	1	1	0	0	2	1	4	1	4
C02	3	1	3	2	1	3	1	1	2	2	2	0	4
C03	0	1	1	3	0	4	3	3	0	2	4	4	4
C04	1	2	1	1	3	0	1	0	3	3	0	1	1
C05	3	3	2	0	1	2	0	3	2	4	4	2	4
C06	3	1	2	3	0	1	2	2	0	3	0	1	2
C07	1	1	3	3	4	4	1	3	2	4	3	3	3
C08	0	0	0	2	4	3	1	1	4	0	0	1	2
C09	3	1	4	4	1	0	3	1	1	4	1	3	4
C10	2	3	4	1	1	3	2	2	3	1	3	2	3
C11	0	1	2	4	2	2	3	2	0	4	1	4	0
C12	4	1	2	1	3	2	1	0	2	3	0	1	0
C13	1	2	0	0	2	0	4	3	1	4	2	3	1

Data availability

Data will be made available on request.

References

Ana Esteso, M.M.E. Alemany, Ángel Ortiz, Impact of product perishability on agrifood supply chains design, Appl. Math. Model. 96 (2021) 20–38.

J. Kandasamy et al.

- [2] Jingying Ding, Zhenkang Peng, Heuristics for perishable inventory systems under mixture issuance policies, Omega 126 (2024) 103078.
- [3] Dos Santos Formiga Anderson, Vivaldo Silveira Júnior, Effects of gaseous hydrogen peroxide on quality, shelf life and distribution management strategy FEFO (first expired, first out) of guavas cv. Pedro Sato, Food Control 165 (2024) 110677.
- [4] Arzum Akkas, Dorothee Honhon, Determining maximum shipping age requirements for shelf life and food waste management, Prod. Oper. Manag. 32 (7) (2023) 2173–2188.
- [5] Navid Mohamadi, Seyed Taghi Akhavan Niaki, Mahdi Taher, Shavandi Ali, An application of deep reinforcement learning and vendor-managed inventory in perishable supply chain management, Eng. Appl. Artif. Intell. 127 (2024) 107403.
- [6] Magdalena Leithner, Christian Fikar, A simulation model to investigate impacts of facilitating quality data within organic fresh food supply chains, Ann. Oper. Res. (2022) 1–22.
- [7] Dileep Kumar Murala, Komali Siddamshetti, Dr Mariofanna Milanova, Leveraging IOT in supply chain sustainability: a provenance perspective. Blockchain, IoT, and AI Technologies for Supply Chain Management: Apply Emerging Technologies to Address and Improve Supply Chain Management, 2024, pp. 147–176.
- [8] Hasnain Abbas, Lindu Zhao, Xi Gong, Narmeen Faiz, The perishable products case to achieve sustainable food quality and safety goals implementing on-field sustainable supply chain model, Soc. Econ. Plann. Sci. 87 (2023) 101562.
- [9] Karam Sallam, Mona Mohamed, Wagdy Mohamed Ali, Internet of Things (IoT) in supply chain management: challenges, opportunities, and best practices, Sustain. Mach. Intell. J. 2 (2023), 3-1.
- [10] Ananya Bhattacharya, Sajad Fayezi, Ameliorating food loss and waste in the supply chain through multi-stakeholder collaboration, Ind. Mark. Manag. 93 (2021) 328–343.
- [11] Emmanuel Adeyemi Abaku, Tolulope Esther Edunjobi, Agnes Clare Odimarha, Theoretical approaches to AI in supply chain optimization: pathways to efficiency and resilience, Int. J. Sci. Technol. Res. Arch. 6 (1) (2024) 92–107.
- [12] Carlos Anfbal Suárez, Walter A. Guaño, Cinthia C. Pérez, Heydi Roa-López, Multiobjective optimization for perishable product dispatch in a FEFO system for a food bank single warehouse, Oper. Res. Perspect. 12 (2024) 100304.
- [13] Yan Bai, Robel Alemu, Steven A. Block, Derek Headey, William A. Masters, Cost and affordability of nutritious diets at retail prices: evidence from 177 countries, Food Policy 99 (2021) 101983.
- [14] Stevens Kisaka, Frank K. Tumwebaze, Expiry of veterinary medicines in supply outlets in Central Uganda: prevalence, management and associated factors, Journal of Pharmaceutical Policy and Practice 16 (1) (2023) 63.
- [15] Ahmed M. Khedr, Enhancing supply chain management with deep learning and machine learning techniques: a review, J Open Innov. Technol. Mark. Complex. (2024) 100379.
- [16] Abhishek Kashyap, Dharmender Yadav, Om Ji Shukla, Rupesh Kumar, Unravelling barriers to food loss and waste in perishable food supply chain: a way toward sustainability, Environ. Dev. Sustain. (2023) 1–21.
- [17] Nasir Ababulgu, Nugusa Abajobir, Hika Wana, The embarking of COVID19 and the perishable products' value chain in Ethiopia, J. Innovat. Entrepren. 11 (1) (2022) 34.
- [18] Javid Jouzdani, Kannan Govindan, On the sustainable perishable food supply chain network design: a dairy products case to achieve sustainable development goals, J. Clean. Prod. 278 (2021) 123060.
- [19] Qasem Gailan, Faisal Aqlan Ahmed, Abdulrahman Shamsan, Mohammed Alhendi, A simulation-optimisation approach for production control strategies in perishable food supply chains, J. Simulat. 17 (2) (2023) 211–227.
- [20] Lena Riesenegger, Maria João Santos, Manuel Ostermeier, Sara Martins, Pedro Amorim, Alexander Hübner, Minimizing food waste in grocery store operations: literature review and research agenda, Sustain. Anal. Model. 3 (2023) 100023.
- [21] Beesetti Lohita, Mupparapu Srijaya, Novel technologies for shelf-life extension of food products as a competitive advantage: a review. Food Production, Diversity, and Safety Under Climate Change, 2024, pp. 285–306.
- [22] Francis Lwesya, Jyoti Achanta, Mapping research trends on food supply chain: a bibliometric analysis, J. Agribus. Dev. Emerg. Econ. 14 (3) (2024) 496–518.
- [23] Anish Kumar, Sachin Kumar Mangla, Pradeep Kumar, An integrated literature review on sustainable food supply chains: exploring research themes and future directions, Sci. Total Environ. 821 (2022) 153411.
- [24] Baha M. Mohsen, Developments of digital technologies related to supply chain management, Procedia Comput. Sci. 220 (2023) 788–795.
- [25] Garza-Reyes, Sudhanshu Joshi, Enhancing resiliency of perishable product supply chains in the context of the COVID-19 outbreak, Int. J. Logist. Res. Appl. 25 (9) (2022) 1219–1243.
- [26] Hafiz Wasim Akram, Samreen Akhtar, Alam Ahmad, Imran Anwar, Mohammad Ali Bait Ali Sulaiman, Developing a conceptual framework model for effective perishable food cold-supply-chain management based on structured literature review, Sustainability 15 (6) (2023) 4907.
- [27] Charles B. Herron, Laura J. Garner, Aftab Siddique, Tung-Shi Huang, Jesse C. Campbell, Shashank Rao, Amit Morey, Building 'First Expire, First Out' models to predict food losses at retail due to cold chain disruption in the last mile, Front. Sustain. Food Syst. 6 (2022) 1018807.
- [28] Mahyar Abbasian, Zeinab Sazvar, Mohammadhossein Mohammadisiahroudi, A hybrid optimization method to design a sustainable resilient supply chain in a perishable food industry, Environ. Sci. Pollut. Control Ser. 30 (3) (2023) 6080–6103.
- [29] Ashish Kumar, Sunil Agrawal, Challenges and opportunities for agri-fresh food supply chain management in India, Comput. Electron. Agric. 212 (2023) 108161.

Journal of Agriculture and Food Research 21 (2025) 101848

- [30] Quan Zhu, Harold Krikke, Managing a sustainable and resilient perishable food supply chain (PFSC) after an outbreak, Sustainability 12 (12) (2020) 5004.
- [31] Tarek Ben Hassen, Hamid El Bilali, Mohammad S. Allahyari, Sinisa Berjan, Oksana Fotina, Food purchase and eating behavior during the COVID-19 pandemic: a cross-sectional survey of Russian adults, Appetite 165 (2021) 105309.
- [32] Junnian Song, et al., Shared network and supply chain features for synergetic control of carbon and air pollutant emissions, Sci. Total Environ. 827 (2022) 154391.
- [33] Kathryn E. Bender, Aishwarya Badiger, Brian E. Roe, Yiheng Shu, Danyi Qi, Consumer behaviour during the COVID-19 pandemic: an analysis of food purchasing and management behaviours in US households through the lens of food system resilience, Soc. Econ. Plann. Sci. 82 (2022) 101107.
- [34] Yang Yu, Edward C. Jaenicke, The effect of sell-by dates on purchase volume and food waste, Food Policy 98 (2021) 101879.
- [35] S.S. Chen, B. Choubey, V. Singh, A neural network-based price sensitive recommender model to predict customer choices based on price effect, J. Retailing Consum. Serv. 61 (2021) 102573.
- [36] Nadia Pourmohammad-Zia, A review of the research developments on inventory management of growing items, J. Supply Chain Manag. Sci. 2 (3–4) (2021) 71–84.
- [37] Berdymyrat Ovezmyradov, Product availability and stockpiling in times of pandemic: causes of supply chain disruptions and preventive measures in retailing, Ann. Oper. Res. (2022) 1–33.
- [38] Christian Fikar, Andreas Mild, Martin Waitz, Facilitating consumer preferences and product shelf life data in the design of e-grocery deliveries, Eur. J. Oper. Res. 294 (3) (2021) 976–986.
- [39] Mahnaz Sohrabi, Mostafa Zandieh, Behrouz Afshar-Nadjafi, A simple empirical inventory model for managing the processed corneal tissue equitably in hospitals with demand differentiation, Comput. Appl. Math. 40 (8) (2021) 281.
- [40] Azmeraw Bekele, Wondwosen Kumsa, Mihret Ayalew, Assessment of inventory management practice and associated challenges of maternal, newborn, and child health life-saving drugs in public hospitals of Southwest Ethiopia: a mixed-method approach, Integrated Pharm. Res. Pract. (2022) 139–149.
- [41] A. Najlae, A. Sedqui, A. Lyhyaoui, A product-driven system to facilitate FEFO application in warehouses, Procedia Comput. Sci. 191 (2021) 451–456.
- [42] Sina Abbasi, Choukolaei Hassan Ahmadi, A systematic review of green supply chain network design literature focusing on carbon policy, Decis. Anal. J. 6 (2023) 100189.
- [43] Mohd Javaid, Abid Haleem, Ravi Pratap Singh, Sumit Gupta, Leveraging lean 4.0 technologies in healthcare: an exploration of its applications, Adv. Biomark. Sci. Technol. 6 (2024) 138–151.
- [44] Imen Mejri, Safa Bhar Layeb, Najla Aissaoui, Mohamed Yassine Kaabar, Inventory management performance of medical analysis laboratory in a public hospital, in: Hospital Supply Chain: Challenges and Opportunities for Improving Healthcare, Springer Nature Switzerland, Cham, 2024, pp. 267–283.
- [45] Thomas K. Dasaklis, Giannis T. Tsoulfas, The future of healthcare supply chains: integrating industry 4.0 technologies for improved resilience and sustainability, in: Hospital Supply Chain: Challenges and Opportunities for Improving Healthcare, Springer Nature Switzerland, Cham, 2024, pp. 533–551.
- [46] V. Polotski, A. Gharbi, J.-P. Kenne, Production control in manufacturing systems with perishable products under periodic demand, J. Manuf. Syst. 63 (2022) 288–303.
- [47] Adaku Edith Adedeji, Fresh Food Products Inventory Control Management: the Challenges in Avoiding Perishability at the Retailers' Shelf, 2020.
- [48] Ole Hansen, Sandra Transchel, Hanno Friedrich, Replenishment strategies for lost sales inventory systems of perishables under demand and lead time uncertainty, Eur. J. Oper. Res. 308 (2) (2023) 661–675.
- [49] Heesup Han, Consumer behavior and environmental sustainability in tourism and hospitality: a review of theories, concepts, and latest research. Sustainable consumer behaviour and the environment, 2021, pp. 1–22.
- [50] Kosuke Motoki, Toshiki Saito, Takuya Onuma, Eye-tracking research on sensory and consumer science: a review, pitfalls and future directions, Food Res. Int. 145 (2021) 110389.
- [51] R.P. Rooderkerk, N. DeHoratius, A. Musalem, Retail analytics: the quest for actionable insights from big data on consumer behavior and operational execution, Chicago Booth Res. Paper 21 (2021) 2110–2139.
- [52] Diriba Feyisa, et al., Adherence to WHO vaccine storage codes and vaccine cold chain management practices at primary healthcare facilities in Dalocha District of Silt'e Zone, Ethiopia, Trop. Dis. Travel Med. Vaccines 8 (1) (2022).
- [53] Otuto Amarauche Chukwu, Maxwell Adibe, Quality assessment of cold chain storage facilities for regulatory and quality management compliance in a developing country context, Int. J. Health Plann. Manag. 37 (2) (2022) 930–943.
- [54] Johann Harer, Jörg Schröttner, Quality management requirements in compliance with European regulations, in: Medical Devices and in Vitro Diagnostics: Requirements in Europe, Springer International Publishing, Cham, 2023, pp. 63–87.
- [55] Abdul Rahman, Noorul Shaiful Fitri, et al., Decision analysis of warehouse productivity performance indicators to enhance logistics operational efficiency, Int. J. Prod. Perform. Manag. 72 (2023) 4.
- [56] Ravi Sankar Pasupuleti, Deepthi Thiyyagura, An empirical evidence on the continuance and recommendation intention of ChatGPT among higher education students in India: an extended technology continuance theory, Educ. Inf. Technol. 29 (14) (2024) 17965–17985.
- [57] Oliyad Kebede, Gizachew Tilahun, Desalegn Feyissa, Storage management and wastage of reproductive health medicines and associated challenges in west Wollega zone of Ethiopia: a mixed cross-sectional descriptive study, BMC Health Serv. Res. 21 (2021) 1–11.

J. Kandasamy et al.

- [58] Ole Hansen, Sandra Transchel, Hanno Friedrich, Replenishment strategies for lost sales inventory systems of perishables under demand and lead time uncertainty, Eur. J. Oper. Res. 308 (2) (2023) 661–675.
- [59] Imen Mejri, Safa Bhar Layeb, Najla Aissaoui, Mohamed Yassine Kaabar, Inventory management performance of medical analysis laboratory in a public hospital, in: Hospital Supply Chain: Challenges and Opportunities for Improving Healthcare, Springer Nature Switzerland, Cham, 2024, pp. 267–283.
- [60] Caterina Trevisan, Marco Formentini, Digital technologies for food loss and waste prevention and reduction in agri-food supply chains: a systematic literature review and research agenda, IEEE Trans. Eng. Manag. 71 (2023) 12326 - 12345.
- [61] Vivek Agrawal, Nitin Seth, Jitendra Kumar Dixit, A combined AHP-TOPSIS-DEMATEL approach for evaluating success factors of e-service quality: an experience from Indian banking industry, Electron. Commer. Res. (2022) 1–33.